Stable Sharing *

Antonio Nicolò[†], Pietro Salmaso[‡], Arunava Sen[§], Sonal Yadav[¶] April 9, 2023

Abstract

We propose a simple model in which agents are matched in pairs in order to complete a task of unit size. The preferences of agents are single-peaked and continuous on the amount of time they devote to it. Our model combines features of two models: assignment games (Shapley and Shubik (1971)) and the division problem (Sprumont (1991)). We provide an algorithm (Select-Allocate-Match) that generates a stable and Pareto efficient allocation. We show that stable allocations may fail to exist if either the single-peakedness or the continuity assumption fail.

JEL classification: C78; D47; D71

Keywords: Job sharing; Matching; Stability; Pareto efficiency

1 INTRODUCTION

Consider a setting where an organization such as a firm, a governmental agency or an NGO, is required to perform multiple tasks. Each task has to be performed by a team consisting of a pair of agents.¹ In order for a task to be completed successfully, the paired agents

^{*}We would like to thank Siddarth Chatterjee, Bhaskar Dutta, Umut Dur, Matthew Jackson, George Mailath, Debasis Mishra, Thayer Morrill, Debraj Ray, Hans Peters, Jay Sethuraman, Rajiv Vohra, Rakesh Vohra and Leeat Yariv for their valuable comments. We thank the reviewers, Advisory Editor and Sushil Bikhchandani for their comments. This paper has also benefited from the comments of various seminar and conference participants. Sonal Yadav gratefully acknowledges financial support from the Jan Wallander and Tom Hedelius Foundation.

[†]Department of Economics, University of Padova, via del Santo 33, 35123 Padova, Italy and School of Social Sciences, The University of Manchester, Manchester M13 9PL, United Kingdom.

[‡]Department of Economics, University of Padova, via del Santo 33, 35123 Padova, Italy

[§]Economics and Planning Unit, Indian Statistical Institute, 7 S.J.S Sansanwal Marg, New Delhi-110016, India.

[¶]Umeå University, Umeå, Sweden.

¹We will also consider the multi-agent team problem briefly in Section 7.

must jointly supply effort levels that sum to one. Each agent dislikes effort. However, they receive "credit" (or equivalently, acquire a reputation) for their work which depends on their observable effort level. The payoff to agent *i* for working an amount t_i is given by $U_i(t_i) = a_i c_i(t_i) - e_i t_i$ where $c_i(\cdot)$ is a strictly concave function that specifies the credit received by *i* as a function of her effort level, a_i is *i*'s valuation of credit and e_i is *i*'s per-unit disutility of effort. A slightly different but equivalent formulation is one where each agent is paid according to a fixed wage rate but has a strictly convex cost of effort, i.e. *i*'s payoff for working is $V_i(t_i) = wt_i - h_i(t_i)$ where *w* is the wage rate and $h_i(\cdot)$ is the strictly increasing convex cost of effort function. An important feature of both models is that agent *i* has an induced preference ordering over effort levels that is single-peaked.

There are several other situations where our model is applicable. Roommates who are assigned together have to share household responsibilities. Some agents clearly prefer to spend more time in activities such as cooking than others and it is reasonable to assume that preferences over time spent on these duties is single-peaked. Pairs of police officers on patrol duty have to share driving and other duties. Sales agents often work in teams to serve a common pool of customers and typically have different preferences over the proportion of customers they would like to serve. In each case, the assumption of single-peaked preferences is natural. All the situations we have described have a sharing problem embedded in a matching problem. Our goal in this paper is to investigate this problem from the perspectives of stability and efficiency.

A pair of agents can block an allocation by proposing contributions summing to one that make them strictly better off than they were in the allocation. An allocation is stable if it cannot be blocked by any pair of agents. Stability is an important requirement in decentralized market design because it can be interpreted as a form of envy-freeness. The motivation for Pareto efficiency is, of course, evident.

Our main result is that a stable and Pareto efficient allocation exists for every profile of single-peaked and continuous preferences. Our existence proof is constructive - we provide an algorithm, the Select-Allocate-Match (SAM) algorithm that identifies a stable and Pareto efficient allocation at every preference profile. Stability fails if either of the assumptions on preferences, single-peakedness and continuity are violated.

The SAM algorithm proceeds by partitioning agents into a set of high type agents (denoted by H) and a set of low type agents (denoted by L) depending upon whether their peaks are greater than or less than 0.5. The algorithm relies on the key notion of an *improvement set* which is defined with respect to an agent's contribution. Roughly speaking, the improvement set for a low type agent is the set of her contributions in the interval [0, 0.5]that would make her strictly better-off. The improvement set for a high type agent is the set of contributions of her partner in [0, 0.5] that would make her strictly better-off. An important result that undergirds our algorithm is that an allocation is stable if the following two conditions are satisfied: (i) the intersection of the improvement sets for each pair of low and high type agents is empty and (ii) 0.5 does not belong to the improvement set of any agent.

In an initializing step, an equal pool of high and low type agents is created by removing the "excess" agents of one type. This is done by choosing agents whose "equivalent contribution" to 0.5 is closest to 0.5.² These agents are matched to each other with each contributing 0.5. In subsequent steps, an agent of one type called the primary agent is selected and her contribution chosen. This is done in a manner such that the primary agent does not want to block with any of the agents who have been assigned in a previous step, by considering their improvement sets. This may involve the primary agent being given a contribution equal to her peak. Then an agent of the opposite type called the secondary agent is selected based on the equivalent of the chosen contribution. We continue in this fashion till all agents are matched. It is worth emphasizing that the partitioning of agents into high and low types is artificial in the sense that blocking pairs can be formed by agents of the same type. In our procedure, we first *select* a primary agent, *allocate* a contribution to her and *match* her with an appropriate secondary agent. This motivates our use of the SAM term. The SAM algorithm works in polynomial time and its time complexity is $O(n^2)$ where n is the number of the agents.

The apparent simplicity of our model may suggest that more naive approaches to finding stable and Pareto efficient allocations exist. The examples in Section 3 show that several natural procedures fail. These include procedures based on rewarding agents on "the short side of the market", "top peak-bottom peak" matching and on the uniform rule (Sprumont (1991)).

The existence of a stable allocation in our problem is far from obvious for several reasons. Unlike the celebrated Shapley-Shubik assignment game (Shapley and Shubik (1971)), we do not have a bipartite structure on the set of agents. Instead, we have a roommate type problem (Gale and Shapley (1962)) where every pair of agents can potentially be matched. It is well-known that non-fractional stable matchings in the roommate model do not exist in general (see Teo and Sethuraman (1998) and Eriksson and Karlander (2001)). There are two additional features of our model that are absent in the Shapley-Shubik model. The first is that the free-disposal assumption is violated in ours. Agents cannot always be made better-off by giving them "more" - making an agent contribute more than her peak results in the agent being worse-off. The second feature is that the assumption of single-peaked preferences implies that our model cannot be represented by a transferable utility game. The underlying non-transferable utility game is hard to analyze using standard techniques because of satiation in preferences, the roommate type structure and the non-convexity of stable allocations (see Section 7.5).

²For an L agent, this is the contribution in [0, 0.5] that is indifferent to 0.5. The equivalent for a H type agent can be suitably defined. The notion of an equivalent contribution is used extensively in the algorithm. We are glossing over important details here which can be found in Section 4.

Our model is also related to the class of division problems first studied in Sprumont (1991). The uniform rule plays a central role in this setting and has been a characterized in a variety of ways (see Sprumont (1991), Sönmez (1994), Ching (1994), Schummer and Thomson (1997) etc.). This model and the uniform rule have been generalized in a variety of ways, for example in Barberà et al. (1997), Bochet et al. (2012), Bochet et al. (2013) and Klaus et al. (1998). However it does not appear to be important for the analysis of stability in our model. We provide an example where there is no agent matching that generates a stable allocation if the uniform rule or its asymmetric variants are used to determine the contribution of the paired agents.³

In our model, agents do not have preferences over their partner as in the classical roommate problem. Nicolò et al. (2019) study a model where agents are matched in pairs and have preferences over both the partner and the project they are assigned to. They show that the existence of stable allocations cannot be guaranteed except when specific assumptions are made on an agent's ranking of partners and projects. The assumption that agents only care about the amount of time (or effort) they devote to the task, and not about their partner or the specific portion of the day or the week they work, is indeed a simplification.⁴

An interesting open question is whether the existence of stable allocations in our model can be derived from existing stability results such as Scarf (1967) and Shapley and Vohra (1991). Proving balancedness of the game and dealing with the absence of free-disposal appears to be non-trivial. In any case, we believe that our approach is more direct and illuminating since we provide an algorithm which generates a stable and Pareto efficient allocation.

The rest of the paper is organized as follows. In Section 2 we introduce the model and basic definitions. Section 3 contains some illustrative examples. Section 4 introduces the concept of improvement sets while Section 6 presents the algorithm and the main result. Section 7 discusses various aspects of our model and results. The proof of stability is contained in Appendix A. The proof of Pareto efficiency is contained in Appendix B.

2 The Model

The set of agents is $N = \{1, ..., n\}$ where n is even. Agents have to be assigned in pairs and each pair has to complete a task of unit value. No agent can remain on her own⁵ and each agent has only one partner.

 $^{^{3}}$ A more detailed discussion of the literature on stability in division problems can be found in Section 7.3.

⁴ The model nevertheless captures relevant features of job sharing such as the demand for reduced working time and the need to find a compatible match. The assumption that workers are indifferent towards their matched partner is likely to be satisfied in routine jobs, or jobs in which the contribution of each worker is fully verifiable.

⁵For further discussion on this assumption, see Section 5.3.

An allocation σ is a collection of triples, (i, j, t_i) where $i, j \in N, t_i \in [0, 1]$ and each agent belongs to exactly one triple. We interpret t_i as the contribution of agent i. The contribution of agent i's partner j is $t_j = 1 - t_i$. We refer to (t_i, t_j) as the contribution vector associated with the matched pair (i, j). We say $(i, j, t_i) \in \sigma$ if the pair (i, j) has the contribution vector (t_i, t_j) in σ . Let Σ denote the set of all feasible allocations.

Each agent *i* has a preference ordering \succeq_i over her contribution.⁶ We assume \succeq_i is singlepeaked and continuous. The ordering \succeq_i is single-peaked if there exists a unique contribution $p_i \in [0,1]$ such that for all $x, y \in [0,1]$, if $x < y \leq p_i$ or $x > y \geq p_i$ then $y \succ_i x$.⁷ The contribution p_i will be referred to as the peak of agent *i* in \succeq_i . A special instance of a single-peaked preference is a symmetric or Euclidean preference: $x \succeq_i y$ if and only if $|x - p_i| \leq |y - p_i|$. The ordering \succeq_i is continuous if the sets $\{y : y \succeq_i x\}$ and $\{y : x \succeq_i y\}$ are closed for all $x \in [0, 1]$. A preference profile \succeq is an *n*-tuple of preferences $(\succeq_1, \ldots, \succeq_n)$.

The fundamental property that an allocation should satisfy is *stability*.

DEFINITION 1 Fix a preference profile \succeq . Let σ be an allocation and $i, j \in N$ be agents with contributions t_i and t_j respectively in σ . Then the pair (i, j) blocks σ at \succeq if there exists a contribution vector (t'_i, t'_j) with $t'_i + t'_j = 1$, $t'_i \succ_i t_i$ and $t'_j \succ_j t_j$. An allocation is stable at \succeq if it cannot be blocked by any pair of agents.

Blocking can occur in two ways. It is possible that i and j are matched together in σ , but can propose an alternative contribution vector which makes both of them better-off.⁸ The other possibility is that i and j are not matched together in σ , but can abandon their partners and come together with a contribution vector which makes both better-off.

A more permissive notion of blocking is weak blocking where only one of the blocking agents is better-off and the other one no worse-off.

DEFINITION 2 Fix a preference profile \succeq . Let σ be an allocation and $i, j \in N$ be agents with contributions t_i and t_j respectively in σ . Then the pair (i, j) weakly blocks σ at \succeq if there exists a contribution vector (t'_i, t'_j) with $t'_i + t'_j = 1$, $t'_i \succeq_i t_i$ and $t'_j \succeq_j t_j$ with either $t'_i \succ_i t_i$ or $t'_j \succ_j t_j$. An allocation is strongly stable at \succeq if it cannot be weakly blocked by any pair of agents.

We are also interested in *Pareto efficient* allocations. Note that we are using the stronger notion of Pareto efficiency.

⁶The asymmetric and symmetric components of \succeq_i are denoted by \succ_i and \sim_i respectively.

⁷The notion of single-peaked preferences is standard - see Mas-Colell et al. (1995). It is used extensively in a variety of contexts such as political economy and axiomatic allocation theory.

⁸See Appendix B for a discussion of blocking by a pair of agents who are matched together in an allocation.

DEFINITION **3** Let σ be an allocation where the contribution of agent i is t_i^{σ} . The allocation τ Pareto dominates σ at preference profile \succeq if $t_i^{\tau} \succeq_i t_i^{\sigma}$ for all $i \in N$ and $t_i^{\tau} \succ_i t_i^{\sigma}$ for some $i \in N$. The allocation σ is Pareto efficient at \succeq if there does not exist $\tau \in \Sigma$ that Pareto dominates it.

Stability and Pareto efficiency are independent properties in our model. Consider a problem with four agents all of whom have symmetric preferences with their peak at 0.3. An allocation where one agent in each pair receives her peak, is Pareto efficient. However it is not stable because their partners contribute 0.7 and can strictly improve by forming a pair with each contributing 0.5.

To show that stability does not imply Pareto efficiency, consider the problem with four agents 1, 2, 3 and 4 who have symmetric preferences with peaks 0.1, 0.2, 0.8 and 0.9 respectively. The allocation (1,3,0.1) and (2,4,0.2) is stable because 1 and 2 are receiving their peaks. It is not Pareto efficient because the allocation (1,4,0.1) and (2,3,0.2) dominates it.

A characterization of stable allocations is not straightforward. However the two propositions below identify some of their important features.

Fix a preference profile \succeq . Partition the set of agents into high type and low type agents depending upon whether their peaks are greater than or equal to or less than 0.5. Formally, $H = \{i \in N : p_i \ge 0.5\}$ and $L = \{i \in N : p_i < 0.5\}$. Furthermore, the set of strictly high type agents is $\hat{H} = \{i \in N : p_i > 0.5\}$.⁹ A mixed pair in an allocation is a pair consisting of an agent from L and an agent from \hat{H} . The next proposition shows that the number of mixed pairs in any stable allocation is maximal.

PROPOSITION 1 In any stable allocation, the number of mixed pairs must be equal to $\min\{|\hat{H}|, |L|\}.$

Proof: Assume for contradiction that there exists an agent $i_1 \in L$ who is matched to an agent $i_2 \notin \hat{H}$ and an agent $j_1 \in \hat{H}$ who is matched to an agent $j_2 \notin L$. There are two cases to consider.

The first case is when each of the agents i_1, i_2, j_1, j_2 contribute 0.5 in σ . There exists $\epsilon > 0$ small enough such that $p_{i_1} \le 0.5 - \epsilon$ and $p_{j_1} \ge 0.5 + \epsilon$. Thus the pair (i_1, j_1) blocks σ with $(0.5 - \epsilon, 0.5 + \epsilon)$.

Suppose the first case does not hold. Then there exists at least one agent whose contribution is not 0.5. Suppose i_1 is one of these agents. We must also have $t_{i_2} \neq 0.5$. Clearly either t_{i_1} or t_{i_2} is greater than 0.5. Suppose $t_{i_2} > 0.5$. Consider the pair (j_1, j_2) . There are two possibilities: $t_{j_1} \leq 0.5$ and $t_{j_1} > 0.5$. If $t_{j_1} \leq 0.5$, then there exists $\epsilon > 0$ small enough such that $0.5 + \epsilon \leq p_{j_1}$ and $0.5 - \epsilon \succ_{i_2} t_{i_2}$. Thus σ is blocked by (i_2, j_1) with $(0.5 - \epsilon, 0.5 + \epsilon)$. If $t_{j_1} > 0.5$, then $t_{j_2} < 0.5$ and the pair (i_2, j_2) blocks σ with (0.5, 0.5).

⁹Note that agents with peak 0.5 are included in the set H. The sets H and L are used throughout the paper except in Propositions 1 and 2 where the sets \hat{H} and L are used.

The remaining case is where i_1 and i_2 contribute 0.5 in σ . Then j_1, j_2 have contributions not equal to 0.5. This case can be dealt with in a manner similar to the earlier case.

Suppose the difference in the cardinalities of the sets \hat{H} and L exceeds two. According to Proposition 2, the surplus agents who do not belong to mixed pairs must contribute 0.5 in any stable allocation.

PROPOSITION 2 In any stable allocation, the following must hold:

- (a). If $|\hat{H}| |L| > 2$, then every agent $i \in \hat{H}$ who is not matched to an agent in L must contribute 0.5.
- (b). If $|L| |\hat{H}| > 2$, then every agent $i \in L$ who is not matched to an agent in \hat{H} must contribute 0.5.

Proof: We only prove Part (a) since the proof of Part (b) is the symmetric analogue. Let σ be a stable allocation. Assume for contradiction that there exists $(i_1, j_1, t_{i_1}) \in \sigma$ where $i_1 \in \hat{H}, j_1 \notin L$ and $t_{i_1} \neq 0.5$. By hypothesis, there exists at least another triple, say $(i_2, j_2, t_{i_2}) \in \sigma$ where $i_2 \in \hat{H}, j_2 \notin L$.

Since $t_{i_1} \neq 0.5$, either $t_{i_1} < 0.5$ or $t_{i_1} > 0.5$ (this implies $t_{j_1} < 0.5$) must hold. Suppose $t_{i_1} < 0.5$. There are two subcases to consider. If $t_{i_2} \leq 0.5$, there exists $\epsilon > 0$ and small enough such that $t_{i_1} < 0.5 - \epsilon$ and $0.5 + \epsilon \succ_{i_2} t_{i_2}$. Then the pair (i_1, i_2) blocks σ with $(0.5 - \epsilon, 0.5 + \epsilon)$. Otherwise, $t_{j_2} < 0.5$. In this case, there exists $\epsilon > 0$ and small enough such that $t_{i_1} < 0.5 - \epsilon$ and $0.5 + \epsilon \succ_{j_2} t_{j_2}$. Then (i_1, j_2) blocks σ with $(0.5 - \epsilon, 0.5 + \epsilon)$.

Suppose $t_{j_1} < 0.5$. Note that $p_{j_1} \ge 0.5$ whereas $p_{i_1} > 0.5$. However, it is easily verified that the argument in the previous paragraph works in this case as well.

3 Illustrative Examples

The purpose of this section is to highlight important features of our model with simple examples. The first example shows that strongly stable allocations may not exist.

EXAMPLE 1 Let $N = \{1, 2, 3, 4\}$. Agents' preferences are symmetric and the peaks are summarized in Table 1.

p_1	p_2	p_3	p_4
0.8	0.3	0.3	0.3

Table 1: Peaks of agents in Example 1.

Consider an arbitrary allocation. Since agents 2, 3 and 4 have identical preferences, we can assume w.l.o.g. that (1, 2) and (3, 4) are the matched pairs. At least one of the agents in $\{3, 4\}$ must have a contribution of at least 0.5. Suppose this agent is 3. The pair (1, 3) blocks with the contribution vector $(t_1, t_3) = (0.8, 0.2)$. Agent 1 is at least as well-off as before while agent 3 is strictly better-off. Clearly there are no strongly stable allocations. \Box

A stable allocation does exist in Example 1, for instance, (1, 2, 0.8) and (3, 4, 0.5). In fact in any stable allocation, agent 1 must receive her peak. Otherwise agent 1 together with the agent who contributes at least 0.5 will block.

Agents whose peaks sum exactly to 1 are obviously perfect matches. This suggests the following procedure for generating stable allocations. Order all pairs of agents by the distance between the sum of their peaks and one, then greedily create pairs of agents using that ordering. Unfortunately this algorithm does not produce a stable matching as the next example shows.

EXAMPLE 2 Let $N = \{1, 2, 3, 4, 5, 6\}$. Agents' preferences are symmetric and the peaks are summarized in Table 2.

p_1	p_2	p_3	p_4	p_5	p_6
0	0.4	0.41	0.42	0.43	0.74

Table 2: Peaks of agents in Example 2

The procedure outlined earlier generates the following matching: (2, 6), (4, 5), (1, 3). By Proposition 2, we know that agents 1, 3, 4, 5 must contribute 0.5 in any stable allocation. Also agent 6 must receives her peak, otherwise (1, 6) can block the allocation with (0.26, 0.74). So agent 2's contribution is 0.26. Then the pair (2, 3) can block with (0.51, 0.49). Thus no stable allocation can be obtained using this procedure. Observe that (1, 6, 0.26), (2, 3, 0.5)(4, 5, 0.5)is a stable allocation.

The following example shows that giving the peaks to either side of the market when the market is balanced (the number of high type agents is equal to the number of low type agents) may not generate a stable allocation. We consider a procedure where agents are partitioned into high and low type agents as before. An allocation is constructed by giving the peaks of the agents on one side of the market and matching them with agents of the other type.

EXAMPLE **3** Let $N = \{1, 2, 3, 4, 5, 6\}$. Agents' preferences are symmetric and the peaks are summarized in Table 3.

p_1	p_2	p_3	p_4	p_5	p_6
0	0.45	0.45	0.65	0.65	0.65

Table 3: Peaks of agents in Example 3

The set of low and high type agents are $\{1, 2, 3\}$ and $\{4, 5, 6\}$ respectively. Assume w.l.o.g. that the pairs in the allocation are (1, 4), (2, 5) and (3, 6). The allocation where all high type agents get their peaks is not stable. Consider the allocation with the triples (1, 4, 0.35), (2, 5, 0.35) and (3, 6, 0.35). The same side pair (2, 3) blocks with the contribution vector (0.5, 0.5). Similarly the allocation where all low type agents get their peaks is not stable. For instance, the allocation (1, 4, 0), (2, 5, 0.45) and (3, 6, 0.45) is blocked by the same side pair (4, 5) using the contribution vector (0.35, 0.65).

Stable allocations exist in this case as well. One such allocation is (1, 4, 0.2), (2, 5, 0.45) and (3, 6, 0.45).

In the model, despite the fact that Proposition 1 suggests a natural partitioning of agents into low and high types, naively following a Shapley and Shubik (1971) type procedure will not work because of the possibility of same-side blocking.

Another "obvious" procedure would be to arrange the agents in order of their peaks from the highest to the lowest. The highest agent would then be matched with the lowest, the second highest with the second lowest and so on. The next example shows that this procedure does not generate stable allocations irrespective of the way contributions are specified.

EXAMPLE 4 Let $N = \{1, 2, 3, 4\}$. The peaks of the agents are summarized in Table 4. Agents 2, 3 and 4 have symmetric preferences while 1 has single-peaked but non-symmetric preferences with the following restriction: $0.35 \sim_1 0.51$.

p_1	p_2	p_3	p_4	
0.39	0.4	0.4	0.9	

Table 4: Peaks of agents in Example 4.

The pairs formed by the procedure are (1, 4) and (2, 3). Let (t_1, t_4) and (t_2, t_3) be their contribution vectors in a stable allocation.

By feasibility, one of the agents $i \in \{2, 3\}$ will have a contribution $t_i \ge 0.5$. Assume w.l.o.g. i = 2. We claim $t_1 \ge 0.35$. If $t_1 < 0.35$, then the pair (1, 2) can block by proposing the contribution vector (0.51, 0.49). Agent 2 strictly improves as $0.49 \succ_2 t_2$. For agent 1, single-peakedness implies $0.35 \succ_1 t_1$. Since $0.35 \sim_1 0.51$, we have $0.51 \succ_1 t_1$ and agent 1 strictly improves. We also claim $t_4 \ge 0.68$. If $t_4 < 0.68$, then (2, 4) can block by proposing (0.31, 0.69). Agent 4 strictly improves by blocking as she moves closer to her peak. For agent 2, symmetry and $t_2 \ge 0.5$ implies $0.31 \succ_2 t_2$. Thus agent 2 also strictly improves.

We have argued that $t_1 \ge 0.35$ and $t_4 \ge 0.68$. Since 1 and 4 are paired together, we have a violation of feasibility. Hence there are no stable allocations with the pairs (1, 4) and (2, 3).

The previous procedure first specified a way to match agents in pairs and then attempted to find suitable contributions. An alternative approach would be to first choose a rule for determining the contributions of agents and then finding a way to form pairs. A natural candidate for such a rule would be one from the class of strategy-proof and Pareto-efficient rules for two agent allotment problems characterized by Moulin (1980) and Barberà et al. (1997). We describe this class below.

Suppose *i* and *j* are matched together. Pick real numbers p_a, p_b with $0 \le p_a \le p_b \le 1$. There exists an agent in the matched pair, say *i* such that her contribution at every profile is given by:¹⁰

$$t_{i} = \begin{cases} p_{i}, & \text{if } p_{a} \leq p_{i} \leq p_{b} \\ \min\{\max\{p_{i}, 1 - p_{j}\}, p_{a}\}, & \text{if } p_{i} < p_{a} \\ \max\{\min\{p_{i}, 1 - p_{j}\}, p_{b}\}, & \text{if } p_{i} > p_{b}. \end{cases}$$
(1)

The uniform rule characterized by Sprumont (1991) is obtained when $p_a = p_b = 0.5$. In this case, the rule is anonymous, i.e. agents are treated symmetrically. On the other hand, the agents are treated asymmetrically whenever $p_a \neq p_b$. For instance, 1 is a dictator if $p_a = 0$ and $p_b = 1$. An important observation pertaining to these rules is that agent *i* either contributes her peak or one of the numbers p_a, p_b .

We consider rules of this type where two arbitrary numbers $(p_a^{i,j}, p_b^{i,j})$ that do not depend on agents' preferences, are assigned to each pair (i, j). Agent *i*'s contribution is computed as in Equation 1. We claim that rules of this type do not generate stable outcomes at every preference profile, irrespective of how the matching is formed.

EXAMPLE 5 Let $N = \{1, 2, ..., 6\}$. Agents' preferences are symmetric and Table 5 summarizes the peaks of the agents.

Here ϵ is chosen to be less than 0.1. Applying Proposition 2 we can conclude that all agents not matched to agent 1 must contribute 0.5. Suppose agent 1 is matched with agent $l \in \{2, 3, 4, 5, 6\}$. In order to prevent 1 and l from blocking with one of the remaining agents, it must be the case that 1 and l must contribute $0.5 - 2\epsilon$ and $0.5 + 2\epsilon$ respectively, i.e. stable allocations are of the form $(i, j, 0.5), (h, k, 0.5), (1, l, 0.5 - 2\epsilon)$ where (i, j, h, k, l) is a permutation of the agents $\{2, 3, 4, 5, 6\}$. According to our earlier observation, the contributions

¹⁰The contribution of j is simply the complement of t_i .

p_1	p_2	p_3	p_4	p_5	p_6
0	$0.5 + \epsilon$				

Table 5: Peaks of agents in Example 5.

of either 1 and l must coincide either with her peak or with one of the numbers $p_a^{1,l}, p_b^{1,l}, l \in \{2, 3, 4, 5, 6\}$ for all ϵ . Observe that the contributions of 1 and l depend on ϵ . Moreover their peaks are distinct from their contributions for all ϵ . Since ϵ can vary continuously while the number of parameters $p_a^{1,l}, p_b^{1,l}$ is finite, it is clearly not possible to find the parameters that generate stable allocations for all profiles.

We have shown that several naive procedures fail. In the following sections, we describe the SAM algorithm which always delivers a stable and Pareto efficient allocation.

4 Improvement sets and stability

We introduce the notion of improvement sets which play a key role in our algorithm.

We partition agents into sets H and L as defined in Section 2. We represent the peaks and the contributions of agents in the interval [0, 0.5]. The peak of a low type agent p_i is represented by $p_i \in [0, 0.5]$. The peak of a high type agent p_i is represented by the point $1 - p_i$ in [0, 0.5].¹¹ We will also represent contributions in [0, 0.5]. Every point in the interval [0.0.5] also represents a contribution vector for a pair of agents. The distance of the point from 0 is the contribution of one agent and the distance of the point from 0.5 is the "excess" over 0.5 of the contribution of the other agent. See Example 6 for an illustration.

Consider agent $i \in L$ with preference \succeq_i (with peak p_i) and contribution t_i . We define the *improvement set* for i at t_i as follows:

$$I_{i,t_i} = \{ x \in [0, 0.5] : x \succ_i t_i \}.$$

Consider agent $i \in H$ with preference \succeq_i (with peak p_i) and contribution t_i . We define the *improvement set* for i at t_i as follows:

$$I_{i,t_i} = \{ x \in [0, 0.5] : 1 - x \succ_i t_i \}.$$

Observe that the improvement set of an agent i, I_{i,t_i} , is empty when $t_i = p_i$.

We make a brief remark about the asymmetry in the definitions of improvement sets for low and high type agents. For an agent $i \in L$, the improvement set consists of contributions

¹¹The peak of a low type agent is measured from left to right starting at 0, while the "excess" of a high type agent is measured from right to left starting from 0.5.



Figure 1: Improvement set for agent 1 in Example 6

in [0, 0.5] which she strictly prefers to t_i . For an agent $i \in H$, the improvement set consists of contributions made by a potential partner in [0, 0.5] which would make agent i strictly better-off relative to t_i .

The assumptions of single-peakedness and continuity on \succeq_i imposes structure on the improvement sets which we record below as an observation.

OBSERVATION 1 The improvement set of an agent is a connected open subset of [0, 0.5] or equivalently an open interval in [0, 0.5].

Example 6 illustrates improvement sets for both low and high type agents.

EXAMPLE 6 Let $N = \{1, 2, 3, 4\}$. Agents' preferences are symmetric. Table 6 summarizes their peaks and contributions. Agents 1 and 2 are matched together as are 3 and 4.

Agent	1	2	3	4
p_i	0.6	0.75	0.10	0.75
t_i	0.3	0.7	0.25	0.75

Table 6: Peaks of agents in Example 6.

Figure 1 shows the improvement set of agent 1 while Figure 2 shows the improvement sets of agents 2 and 3. The improvement set of agent 4 is empty. Improvement sets of high (low) types are indicated above (below) the [0, 0.5] line.

It is useful to define the notion of an *equivalent* contribution. We denote the equivalent contribution for agent i at t_i by $e_i(t_i)$ when $i \in L$ and $e_i(1 - t_i)$ when $i \in H$.¹²

Consider $i \in L$. If there exists a contribution $x \in [0, 0.5]$ such that $x \sim_i t_i$ and $x \neq t_i$, then $e_i(t_i) = x$. The equivalent $e_i(t_i)$ is an end-point of the improvement set but not included in it. Here the improvement set is one of the following: $(t_i, e_i(t_i))$ if $t_i < p_i$, $(e_i(t_i), t_i)$ if $p_i < t_i \leq 0.5$ or $(e_i(t_i), 0.5]$ if $p_i < 0.5 < t_i$.

¹²We suppress the dependence of the equivalent of agent i on \succeq_i since we keep the latter constant throughout the analysis.



Figure 2: Improvement sets in Example 6

If such an x does not exist, then $e_i(t_i)$ is defined as follows:

$$e_i(t_i) = \begin{cases} -\epsilon & \text{if } I_{i,t_i} = [0, t_i) \text{ or } I_{i,t_i} = [0, 0.5] \\ 0.5 + \epsilon & \text{if } I_{i,t_i} = (t_i, 0.5] \\ t_i & \text{if } I_{i,t_i} = \emptyset \end{cases}$$

where ϵ is any small positive number.

In Example 6, 3 is a low type agent with $p_3 = 0.10$ and contribution $t_3 = 0.25$. Since $0 \succ_3 0.25, e_3(t_3) = -\epsilon$. In fact $e_3(t_3) = -\epsilon$ for all $t_3 > 0.25$. If $t_3 = 0.05$ then $e_3(t_3) = 0.15$.

Consider $i \in H$. If there exists a contribution $x \in [0, 0.5]$ such that $1 - x \sim_i t_i$ and $1 - x \neq t_i$, then $e_i(1 - t_i) = x$. As before, the equivalent $e_i(1 - t_i)$ is an end-point of the improvement set but not included in it. Here the improvement set is one of the following: $(1 - t_i, e_i(t_i)), (e_i(t_i), 1 - t_i)$ or $(e_i(t_i), 0.5]$ if $t_i < 0.5$.

If such an x does not exist, then $e_i(1-t_i)$ is defined as follows:

$$e_i(1-t_i) = \begin{cases} -\epsilon & \text{if } I_{i,t_i} = [0, 1-t_i) \text{ or } I_{i,t_i} = [0, 0.5] \\ 0.5 + \epsilon & \text{if } I_{i,t_i} = (1-t_i, 0.5] \\ 1-t_i & \text{if } I_{i,t_i} = \emptyset \end{cases}$$

where ϵ is any small positive number.

In Example 6, 1 is a high type agent with $p_1 = 0.6$ and $t_1 = 0.3$. Observe $0.9 \sim_1 0.3$. Thus there exists $x = 0.1 \in [0, 0.5]$ such that $1 - x \sim_1 0.3$. Consequently $e_1(1 - t_1) = 0.1$. If $t_1 = 1$, the improvement set for agent 1 is (0, 0.5]. Thus $e_1(1 - t_1) = 0.5 + \epsilon$.

Improvement sets provide a natural way to check for the existence of stable allocations. For instance, in Example 6 agents 2 and 3 who receive contributions 0.7 and 0.25 respectively in an allocation will block. This is evident from the fact that their improvement sets have a non-empty intersection.

DEFINITION 4 A allocation satisfies Condition S if the associated improvement sets satisfy the following:

1. For every $h \in H$ and $l \in L$, $I_{h,t_h} \cap I_{l,t_l} = \emptyset$.

2. For all $i \in N$, $0.5 \notin I_{i,t_i}$.

PROPOSITION 3 If an allocation satisfies Condition S, it is stable. Moreover if an allocation is stable, it satisfies Part 1 of Condition S.

Proof: Consider an allocation that satisfies Condition S but is not stable, i.e. there exists a pair of agents who block. There are two cases to consider.

Case 1: The blocking pair is (l, h) where $l \in L$ and $h \in H$. Let t_l and t_h be the contributions of l and h respectively in the allocation. Suppose they block with the contribution vector (t'_l, t'_h) . If $t'_l \geq 0.5$, single-peakedness implies $0.5 \in I_{l,t_l}$ which would contradict Part 2 of Condition S. Therefore $t'_l < 0.5$, i.e. $1 - t'_l = t'_h > 0.5$. Since $t'_l \succ_l t_l$ and $t'_l < 0.5$, we have $t'_l \in I_{l,t_l}$. Since $t'_h \succ_h t_h$ and $t'_h = 1 - t'_l > 0.5$, we have $t'_l \in I_{h,t_h}$. Therefore $t'_l \in I_{l,t_l} \cap I_{h,t_h}$ contradicting Part 1 of Condition S.

Case 2: Both agents in the blocking pair are of the same type. Suppose (l_1, l_2) is the blocking pair where $l_1, l_2 \in L$. Let t_{l_1} and t_{l_2} be the contributions of agents l_1 and l_2 respectively in the allocation. In the contribution vector used to block, at least one of the agents in the pair, say l_1 has a contribution of at least 0.5. Single-peakedness implies $0.5 \in I_{l_1,t_{l_1}}$ contradicting Part 2 of Condition S. The argument in the case where both agents are high type is virtually identical.

We now show that any stable allocation must satisfy Part 1 of Condition S.

Consider a stable allocation. Assume for contradiction that there exist $i \in L$ and $j \in H$ such that $I_{i,t_i} \cap I_{j,t_j} \neq \emptyset$ where t_i and t_j are contributions of agents i and j respectively in the allocation. Consider $x \in I_{i,t_i} \cap I_{j,t_j}$. By the definition of improvement sets, we have $x \succ_i t_i$ and $1 - x \succ_j t_j$. Thus the pair (i, j) blocks the allocation with the contribution vector (x, 1 - x).

Part 2 of the Condition S can be interpreted as a fairness requirement such that no agent is worse off than the case where she equally shares the contribution with her partner. Note that Part 2 of Condition S is not necessary for the existence of stable allocations. Consider the case where there are four agents l_1, l_2, h_1, h_2 with symmetric preferences. The peaks of l_1 and l_2 are 0.4 and 0.3 while the peaks of h_1 and h_2 are 0.9 and 0.7. The allocation with the triples $(l_1, h_1, 0.1)$ and $(l_2, h_2, 0.3)$ is stable because all agents except l_1 are satiated. However $0.5 \in I_{l_1,0.1}$.

5 The SAM Algorithm: A broad overview

The main contribution of this paper is the Select-Allocate-Match (SAM) algorithm that generates a stable and Pareto efficient allocation for every instance of the sharing problem.¹³

¹³For stability, we show that the allocation generated by SAM algorithm satisfies Condition S.

In this section we informally discuss key ideas in the algorithm and illustrate them with examples. For expositional convenience, we shall mainly confine our discussion to the case of symmetric preferences with a few remarks at the end regarding the general single-peaked case.

The SAM algorithm proceeds as follows. In the initial step (Step 0), the number of high and low type agents are counted and "excess agents" from the larger side, identified. The excess agents are arbitrarily paired with each other and assigned contributions of 0.5. The improvement sets of the paired agents are then calculated. Inputs to the algorithm at Step q are the set of agents paired up to (and including) Step q-1 and the improvement sets of these agents. We let D_q and U_q denote the union of the improvement sets of the low and high type agents respectively who have been paired upto (and including) Step q-1. It is clear that D_q and U_q are unions of open intervals. In Step q, a low type agent (say l) and a high type agent (say h) are chosen with the property that their peaks are closest to 0.5 among the set of unmatched low and high type agents respectively. These agents are paired together and their contribution vector decided using the D_q and U_q sets and the peaks p_l , p_h . Three considerations are involved in choosing this contribution vector (i) the contribution of l lies in between $1 - p_h$ and p_l (ii) l's improvement set does not intersect with U_q and (iii) h's improvement set does not intersect with D_q . The manner in which agents are paired ensures that all three requirements can be reconciled. The contribution of agent l belongs to the set $\{p_l, 1 - p_h, \inf D_q, 1 - \inf U_q\}$. An important feature of the algorithm is that contributions vectors are specified in a manner such that 0.5 does not belong to the improvement set of any agent. This ensures that "same-side" blocking does not occur. We illustrate the working of the algorithm in several examples below.

In Example 1, there are two excess low type agents. Since all low type agents are identical, two of them, say 2 and 3 are picked arbitrarily in Step 0. They are paired together with the contribution vector (0.5, 0.5). Their improvement sets are (0.1, 0.5) so that $U_1 = (0.1, 0.5)$. In Step 1, the remaining agents 1 and 4 are paired. If 1's contribution differs from 0.8, her improvement set will be an open interval with 0.2 as the mid-point. Since 0.2 belongs to U_1 , the algorithm fixes 1's contribution as her peak, i.e the triple (1, 4, 0.8) is generated.

In Example 2, there are five low type agents and one high type agent. The excess agents are the four low type agents whose peaks are closest to 0.5, i.e. agents 2, 3, 4 and 5. In Step 0, the algorithm pairs agents (2,3) and (4,5) with each agent contributing 0.5. The improvement sets for agents 2, 3, 4 and 5 are (0.3, 0.5), (0.32, 0.5), (0.33, 0.5) and (0.34, 0.5) respectively. Thus $D_1 = (0.3, 0.5)$ and U_1 is empty. In Step 1, agents 1 and 6 will be paired. Here $1 - p_6 = 0.26 < \inf D_1$. The algorithm forms the triple (1, 6, 0.26) and agent 6 gets her peak. Observe that the requirements (i), (ii) and (iii) stated earlier are satisfied.

EXAMPLE 7 Let $N = \{1, 2, 3, 4, 5, 6\}$. Agents' preferences are symmetric and the peaks are summarized in Table 7.

p_1	p_2	p_3	p_4	p_5	p_6
0.18	0.2	0.45	0.65	0.66	0.9

Table 7: Peaks of agents in Example 7

Consider Example 7 above. Here there are no surplus agents and no agents are paired in Step 0. Thus D_1 and U_1 are empty. In Step 1, the highest low type agent is paired with the lowest high type agent, with the agent with peak closest to 0.5 getting her peak - the triple (3, 4, 0.45) is formed. The improvement sets for agents 3 and 4 are the null set and (0.25, 0.45) respectively. Thus D_2 is empty and $U_2 = (0.25, 0.45)$. In Step 2, agents 2 and 5 are paired. Since $0.2 = p_2 < \inf U_2 < 1 - p_5 = 0.34$, the algorithm forms the triple (2, 5, 0.25) where agent 2's contribution is min{ $\inf U_2, 1 - p_5$ }. None of the agents get their peak in Step 2 and thus have non-empty improvement sets. The improvement sets for agents 2 and 5 are (0.15, 0.25) and (0.25, 0.43). Note that 2's improvement set does not intersect with U_2 , fulfilling requirement (*ii*). Requirement (*iii*) is satisfied trivially since D_2 is empty while (*i*) is satisfied by construction. The inputs into Step 3 are $D_3 = (0.15, 0.25)$ and $U_3 = (0.25, 0.43) \cup U_2 = (0.25, 0.45)$.

In Step 3, the pair (1,6) is formed. Observe that $1 - p_6 < \inf D_3 < p_1 < \inf U_3$. The algorithm forms the triple (1,6,0.15). In particular, the contribution of agent 1 is $\min\{\inf D_3, p_1\}$. The improvement set of agent 1 is (0.15, 0.21) and does not intersect U_3 . The improvement set of agent 6 is (0.05, 0.15) and does not intersect D_3 . Once again, all three requirements are satisfied.

Example 8 is a variant of Example 7 where only the peak of agent 1 has changed.

EXAMPLE 8 Let $N = \{1, 2, 3, 4, 5, 6\}$. Agents' preferences are symmetric and the peaks are summarized in Table 8.

p_1	p_2	p_3	p_4	p_5	p_6
0.05	0.2	0.45	0.65	0.66	0.9

Table 8: Peaks of agents in Example 8

Steps 0, 1 and 2 are identical to their counterparts in Example 7 as all other peaks remain unchanged. In Step 3, the pair (1,6) is formed. Note that $p_1 < 1 - p_6 < \inf D_3 < \inf U_3$. The algorithm forms the triple (1,6,0.1) where the contribution of agent 1 is min{ $1 - p_6$, inf U_3 } and agent 6 gets her peak. The improvement set of agent 1 is (0,0.1) and does not intersect U_3 . Since 6 gets her peak, she does not participate in any blocking coalition.

Example 9 is a variant of Example 8 where only the peak of agent 6 has changed.

EXAMPLE 9 Let $N = \{1, 2, 3, 4, 5, 6\}$. Agents' preferences are symmetric and the peaks are summarized in Table 9.

p_1	p_2	p_3	p_4	p_5	p_6
0.05	0.2	0.45	0.65	0.66	0.8

Table 9: Peaks of agents in Example 9

Steps 0,1 and 2 are identical to their counterparts in Examples 7 and 8. In Step 3, the pair (1,6) is formed. Note that $\inf D_3 = 0.15 < 0.2 = 1 - p_6$ and $1 - p_6$ belongs to D_3 . Thus agent 6 must be given her peak to prevent her from blocking with the already matched low type agents. The algorithm forms the triple (1,6,0.2). The improvement set of agent 1 is $(0.16, 0.2) \subset D_3$ and does not intersect U_3 .¹⁴

A possible issue of concern regarding the algorithm is that requirements (ii) and (iii) for the assignment of contributions in some Step q may be in conflict with each other. In particular, a situation may arise where $p_l \in U_q$ and $1 - p_h \in D_q$. Then both agents l and h must be assigned their peaks which would be impossible if $p_l + p_h \neq 1$. In the proof of Theorem 1, we show that this situation can never occur. For an insight into this claim, consider the following variant of Example 9. Recall that $1 - p_6 \in D_3$. Suppose p_1 is adjusted so that $p_1 \in U_3 = (0.25, 0.45)$. Then $p_2 = 0.2 < 0.25 \leq p_1$. However this cannot occur because the algorithm would then have paired agent 1 (with agent 5) in Step 2, instead of agent 2.

The algorithm works on the same principle for the general single-peaked case. Improvement sets are no longer based on distances from the peak, but still remain open intervals. The selection of low and high type agents to be paired in any step is now more delicate and depends generally on agent preferences, and not simply on the distance of peaks from 0.5. These details are fully specified in Section 6 - here we briefly summarize the general procedure that motivates the name of the algorithm. In every Step $q \ge 1$, we perform the following operations.¹⁵

• Select an agent i amongst the set of unmatched agents. We call this agent as the primary agent. The following three cases¹⁶ identify the primary agent:

¹⁴The fact that the improvement set of agent 1 is a subset of D_3 is not a coincidence. This follows from the fact that the low type agents matched so far have higher peaks than agent 1 and preferences are symmetric.

¹⁵As described earlier, in Step 0, we match in pairs the "excess" agents of one type so that in Step 1 there are an equal number of unmatched high and low type agents.

¹⁶In the Appendix we prove that these cases are mutually exclusive.

- (i) If there is a high type agent whose peak is less than $1 \inf D_q$, then the primary agent is a high type agent. If there are several such high type agents, the primary agent is the one in this set whose peak is closest to 0.5.
- (ii) If there is a low type agent whose peak is greater than $\inf U_q$, then the primary agent is a low type agent. If there are several such low type agents, the primary agent is the one in this set whose peak is closest to 0.5.
- (iii) If neither (i) nor (ii) hold, the primary agent is the agent whose peak is closest to 0.5.
- Allocate the primary agent, her contribution t_i . If the primary agent is a low type agent, she receives the minimum of her peak and $\inf D_q$. If the primary agent is a high type agent, she receives the maximum of her peak and $1 \inf U_q$.
- Match the primary agent with a secondary agent. If the primary agent i is a high type agent, the secondary agent is the low type agent with the highest equivalent of $(1 t_i)$. If the primary agent i is a low type agent, the secondary agent is the high type agent with the highest equivalent of t_i .

In the special case where all agents have symmetric preferences, the primary and secondary agents are the agents of each type amongst the set of unmatched agents whose peaks are closest to 0.5.

6 The Select-Allocate-Match (SAM) Algorithm

In this section, we provide a formal description of our algorithm and state our main result.

In the rest of the paper, we adopt the following convention: whenever we write a triple (i, j, t_i) in the description of an allocation, we assume $p_i \leq p_j$.¹⁷

Let \succ^N be a linear ordering of the set N. This ordering will serve as a tie-breaking rule. Fix an arbitrary preference profile \succeq . The peaks of the agents at \succeq are p_1, p_2, \ldots, p_n . We begin by partitioning the set of agents N into the sets H and L.

Step 0: We remove excess agents from either the set H or the set L to ensure that the cardinality of the adjusted H and L sets is equal. If |H| > |L|, we remove |H| - |L| agents (chosen in a specific way) from the set H. We denote the set of agents removed from H by \overline{H} . Similarly if |L| > |H|, we remove |L| - |H| agents from L. The set of agents removed from L is denoted by \overline{L} . In addition, we define two sets U_1 and D_1 with $U_1, D_1 \subseteq [0, 0.5]$.

There are three possibilities to consider.

¹⁷Suppose agents 1 and 2 are paired in an allocation. Let $p_1 = 0.4$, $p_2 = 0.7$ and their contributions in the allocation be $t_1 = 0.1$, $t_2 = 0.9$. We shall write the triple as (1, 2, 0.1).

- 1. |H| = |L|. Here $\overline{H} = \emptyset$ and $\overline{L} = \emptyset$. Also $U_1 = \emptyset$ and $D_1 = \emptyset$.
- 2. |H| > |L|. Compute $e_i(0.5)$ for all $i \in H$. Pick the |H| |L| agents whose equivalents $e_i(0.5)$ are closest to 0.5. Ties are broken using the ordering \succ^N . The set of these agents is \bar{H} . Pair the agents in \bar{H} arbitrarily and the contribution of all agents is 0.5. Here $\bar{L} = \emptyset$, $U_1 = \bigcup_{i \in \bar{H}} I_{i,0.5}$ and $D_1 = \emptyset$.
- 3. |H| < |L|. Compute $e_i(0.5)$ for all $i \in L$. Pick the |L| |H| agents whose equivalents $e_i(0.5)$ are closest to 0.5. Ties are broken using the ordering \succ^N . The set of these agents is \bar{L} . Pair the agents in \bar{L} arbitrarily and the contribution for all agents is 0.5. Here $\bar{H} = \emptyset$, $D_1 = \bigcup_{i \in \bar{L}} I_{i,0.5}$ and $U_1 = \emptyset$.

The adjusted partition of N is $H_1 = H \setminus \overline{H}$ and $L_1 = L \setminus \overline{L}$. By construction, $|H_1| = |L_1| = K$. The algorithm has K + 1 steps including Step 0. In each Step q (where $1 \leq q \leq K$), we form a pair consisting of a high type agent and a low type agent. We denote these agents by h_q and l_q respectively. At the start of the generic step q where $q \in \{1, \ldots, K\}$, the algorithm is provided three inputs: (i) Preferences of the agents in H_q and L_q (ii) U_q where $U_q \subseteq [0, 0.5]$ and (iii) D_q where $D_q \subseteq [0, 0.5]$. The set D_q is the union of the improvement sets of the L type agents who have been matched till Step q. A similar comment holds for U_q and H type agents.

Step q: Each step q is divided into four substeps, referred to as Substep q.s where $s \in \{1, 2, 3, 4\}$. We will determine the agents $l_q \in L_q$ and $h_q \in H_q$ who will be matched to each other and their contribution vector (t_{l_q}, t_{h_q}) . At the end of the step, we will determine L_{q+1} , H_{q+1} , D_{q+1} , and U_{q+1} . In case D_q or U_q is empty, we adopt the convention that infimum of D_q and U_q is ∞ .

Step q.1: Consider the set $\{h \in H_q : 1 - p_h > \inf D_q\}$. If it is empty, proceed to Step q.2. Otherwise, choose h_q to be the agent with the lowest peak (or the highest $1 - p_h$) in this set. The agent h_q is the primary agent in this substep. The contribution of agent h_q is $t_{h_q} = \max\{p_{h_q}, 1 - \inf U_q\}$. Choose l_q to be the low type agent in L_q who has the highest $e(1 - t_{h_q})$ (using the tie-breaking ordering \succ^N on agents if necessary). The agent l_q is the secondary agent in this substep. We add the triple $(l_q, h_q, 1 - t_{h_q})$ to the allocation and proceed to Step q.4.

Step q.2: Consider the set $\{l \in L_q : p_l > \inf U_q\}$. If it is empty, proceed to Step q.3. Otherwise, choose l_q to be the agent with the highest peak in this set. The contribution of agent l_q is $t_{l_q} = \min\{p_{l_q}, \inf D_q\}$. Choose h_q to be the high type agent in H_q who has the highest $e(t_{l_q})$ (using the tie-breaking ordering \succ^N on agents in case of ties). We add the triple (l_q, h_q, t_{l_q}) to the allocation and proceed to Step q.4. In this substep, l_q is the primary agent while h_q is the secondary agent. Step q.3: If $1 - p_h \leq \inf D_q$ for all $h \in H_q$ and $p_l \leq \inf U_q$ for all $l \in L_q$, we identify the following agents.

- 1. The high type agent with the lowest peak in H_q . Denote this agent by \tilde{h}_q . In case there is more than one such agent, use \succ^N as the tie breaker.
- 2. The low type agent with the highest peak in L_q . Denote this agent by \tilde{l}_q . Use \succ^N as the tie breaker if required.

There are two possibilities leading to Steps q.3.1 and q.3.2.

Step q.3.1: If $p_{\tilde{l}_q} \leq 1 - p_{\tilde{h}_q}$, choose $h_q = \tilde{h}_q$. The contribution of agent h_q is $t_{h_q} = \max\{p_{h_q}, 1 - \inf U_q\}$. Choose l_q to be the low type agent in L_q who has the highest $e(1 - t_{h_q})$ (using the tie-breaking ordering \succ^N on agents if necessary). We add the triple $(l_q, h_q, 1 - t_{h_q})$ to the allocation and proceed to Step q.4. In this substep, h_q is the primary agent while l_q is the secondary agent.

Step q.3.2: If $p_{\tilde{l}_q} > 1 - p_{\tilde{h}_q}$, choose $l_q = \tilde{l}_q$. The contribution of agent l_q is $t_{l_q} = \min\{p_{l_q}, \inf D_q\}$. Choose h_q to be the high type agent in H_q who has the highest $e(t_{l_q})$ (using the tie-breaking ordering \succ^N on agents in case of ties). We add the triple (l_q, h_q, t_{l_q}) to the allocation. Proceed to Step q.4. In this substep, l_q is the primary agent while h_q is the secondary agent.

Step q.4: Sets $D_{q+1} = D_q \cup I_{l_q,t_{l_q}}$ and $U_{q+1} = U_q \cup I_{h_q,t_{h_q}}$. Also sets $H_{q+1} = H_q \setminus \{h_q\}$ and $L_{q+1} = L_q \setminus \{l_q\}$. The set H_{q+1} (or L_{q+1}) contains the high type (or low type) agents who are unmatched after Step q. Proceed to Step q + 1.

٠		
٠		
٠		
•		
•		
٠		
٠		
٠		

Step K: Note that $|L_K| = |H_K| = 1$. After the completion of this step, all agents in N have been matched and the algorithm terminates.

We state our result below.

THEOREM 1 The SAM algorithm generates a stable and Pareto efficient allocation.

The proof of Theorem 1 is in the Appendices. Appendix A contains the proof of stability and Appendix B (Supplementary Appendix) contains the proof of Pareto efficiency. The allocation generated by the SAM algorithm is stable since it satisfies Condition S. The key step in the proof is to show that the sets D_q and U_q do not intersect and do not contain 0.5 at any step q of the algorithm. The proof of Pareto efficiency requires several steps.

7 DISCUSSION

In this section, we discuss various aspects of our model.

7.1 Single-peaked preferences

In Example 10 we show that the single-peakedness assumption on preferences is vital for the existence of stable allocations.

EXAMPLE 10 Let $N = \{1, 2, ..., 6\}$. Table 10 summarizes the peaks of agents 1 to 5 and the dip of agent 6. All agents in $\{1, ..., 5\}$ have symmetric single-peaked preferences. Agent 6 has symmetric single-dipped preferences with 0.5 as the dip. This means that 0.5 is her worst contribution and she is progressively better-off as she moves farther away from 0.5. As a result, 0 and 1 are her most preferred contributions.

Agent	1	2	3	4	5	6
Peak	0.49	0.49	0.49	0.01	0.98	-
Dip	-	-	-	-	-	0.5

Table 10: Peaks/dip of agents in Example 10.

We argue that there are no stable allocations. Notice that one of the agents in $\{1, 2, 3\}$ must be paired with an agent from $\{4, 5, 6\}$. Assume w.l.o.g. that agent 3 is paired with an agent from $\{4, 5, 6\}$. We consider each case in turn.

Case A: Agent 3 is paired with agent 4. Let their contribution vector be (t_3, t_4) . If the allocation is stable, it must be the case that $t_3 \ge 0.49$ and $t_4 \ge 0.01$.¹⁸

One of the agents 3, 4 must be at a distance of $\max\{t_3 - 0.49, t_4 - 0.01\}$ for any (t_3, t_4) . Minimising over (t_3, t_4) , we infer that one of the agents must be a distance of at least 0.25 from her peak. Suppose this agent is 3. An immediate consequence is that agents 1 and 2 must be receiving their peaks in the allocation. Otherwise agent 3 can block with the non-satiated agent by offering her 0.49 and being only 0.02 away from her own peak. This implies that agents 1 and 2 are not paired together but are paired with 5 and 6. Moreover agents 5 and 6 will each get a contribution of 0.51. Then the pair (5,6) blocks with the contribution vector (0.98, 0.02).

Suppose agent 4 is the agent who is at a distance of at least 0.25 from her peak. Then agent 5 must get her peak and agent 6 must be getting either 0 or 1. If not, agent 4 can

¹⁸If $t_3 > 0.49$ and $t_4 < 0.01$, then the pair (3,4) blocks by proposing a contribution vector (t'_3, t'_4) such that $t'_3 < t_3$ and $t'_4 > t_4$. Similarly the case $t_3 < 0.49$ and $t_4 > 0.01$ is ruled out. Of course $t_3 < 0.49$ and $t_4 < 0.01$ is infeasible.

block with 5 by offering her 0.98 and being 0.01 away from her peak. Agent 4 can block with agent 6 by offering her 1 and being 0.01 away from her own peak. It follows that 5 and 6 cannot be paired together but are paired with 1 and 2. The agent paired with 5, say agent 1 gets 0.02. While agent 2 (paired with 6) gets either 0 or 1. In either case, the pair (1, 2) blocks with the vector (0.5, 0.5). These arguments establish that Case A cannot occur.

Case B: Agent 3 is paired with agent 5. Let (t_3, t_5) be the contribution vector. Using arguments similar to those in Case A, we can argue that $t_3 \leq 0.49$ and $t_5 \geq 0.98$. One of the agents 3, 5 must be at a distance of max $\{0.49 - t_3, 0.98 - t_5\}$ for any (t_3, t_5) . Therefore either 3 or 5 must be at a distance of at least 0.235 from her peak.

Suppose agent 3 is this agent. Like in Case A, agents 1 and 2 must receive their peaks. Thus they are not paired together and are paired with 4 and 6. Moreover 4 and 6 are each receiving 0.51. The pair (4, 6) blocks with (0, 1).

Suppose agent 5 is at a distance of at least of 0.235 from her peak. Like in Case A, agents 4 and 6 must get their peaks. So they cannot be paired together and are paired with 1 and 2. Once again, 1 and 2 will form a blocking coalition.

Case C: Agent 3 is paired with agent 6. Since 1, 2 and 3 have the same preferences, we can apply Cases A and B to argue that neither 1 nor 2 can be matched to an agent in $\{4, 5\}$. Consequently the pairs in this allocation are (1, 2) and (4, 5). Note that there must be an agent in each pair who does not get her peak. Assume w.l.o.g. that 1 is not getting her peak. To ensure that (1,3) does not block with (0.49, 0.51), it must be the case that $0.47 \leq t_3 \leq 0.51$. Thus $0.49 \leq t_6 \leq 0.53$. If 4 is the agent in the pair (4,5) who is not getting her peak, then (4,6) blocks with (0.01, 0.99). If 5 is the agent not getting her peak, then (5,6) blocks with (0.98, 0.02).

Therefore Case C cannot occur and there are no stable allocations.

Example 10 illustrates the key role played by the "complementarity of preferences" in the existence of stable allocations in our model. For simplicity, suppose all agents have symmetric (single-peaked) preferences. Consider two agents of very high type (with peaks close to one) and one of a very low type (with a peak close to zero). Each of the two high type agents are a "good fit" for the low type agent but are not well-suited to be paired together. This prevents the cyclical pattern of blocking which typically underlies the non-existence of stable allocations. This is exactly what occurs in Example 10 - agents 4, 5 and 6 are mutually "good fits" for each other.

7.2 Continuity of preferences

The following example shows that stable allocations may not exist if preferences are singlepeaked but not continuous. EXAMPLE 11 Let $N = \{1, 2, ..., 6\}$. Table 11 summarizes the peaks of the agents. Preferences of agents 1 and 2 are symmetric and continuous. For any agent $i \in \{3, 4, 5, 6\}$, \succeq_i is single-peaked but not continuous at 0.3. In particular, \succeq_i satisfies: (i) for any z such that $0.3 < z \le 0.41$, $z \succ_i 0.5$ and (ii) $\exists \bar{\epsilon} > 0$ such that $0.5 + \bar{\epsilon} \succ_i 0.3$. Continuity of \succeq_i will imply $0.3 \succeq_i 0.5$. Single-peakedness implies $0.5 \succ_i 0.5 + \bar{\epsilon}$. Thus $0.3 \succ_i 0.5 + \bar{\epsilon}$ contradicting (ii).

Agent	1	2	3	4	5	6
Peak	0.7	0.7	0.41	0.41	0.41	0.41

Table 11: Peaks of agents in Example 11.

We claim that stable allocations do not exist.

Consider an arbitrary allocation. If it is stable, agents 1 and 2 are not paired together. If they are, one of them (say 1) gets a contribution of at most 0.5. One of the agents in $\{3, 4, 5, 6\}$, say 3 has a contribution of at least 0.5. Then the pair (1,3) blocks with (0.51, 0.49).

We can therefore assume w.l.o.g. that the pairs (1,3), (2,4) and (5,6) belong to the allocation. By feasibility, one of the agents in $\{5,6\}$, say 5 has a contribution $t_5 \ge 0.5$.

Let (t_1, t_3) be the contribution vector for the pair (1, 3). In order for (1, 3) not to block, we must have $0.59 \le t_1 \le 0.7$ and $0.3 \le t_3 \le 0.41$. There are two cases to consider.

The first is when $t_3 > 0.3$. Since $t_3 \leq 0.41$, we can find $\delta > 0$ small enough such that $t_3 - \delta \in (0.3, 0.41)$ and $t_1 + \delta < 0.7$. By assumption (i), $t_3 - \delta \succ_5 t_5$ as $t_5 \geq 0.5$. Therefore the pair (1,5) can block with $(t_1 + \delta, t_3 - \delta)$.

The remaining case is $t_3 = 0.3$. If $t_5 > 0.5$, the pair (3,5) blocks with (0.5,0.5). Agent 3 strictly improves as $0.5 \succ_3 0.3$ (Assumption (*ii*) and single-peakedness). Agent 5 strictly improves as she moves closer to her peak. Suppose $t_5 = 0.5$. Pick $0 < \epsilon < \bar{\epsilon}$ where $\bar{\epsilon}$ is specified in Assumption (ii). By single-peakedness and Assumption (ii), $0.5 + \epsilon \succ_3 0.5 + \bar{\epsilon} \succ_3$ 0.3. Hence (3,5) blocks with $(0.5 + \epsilon, 0.5 - \epsilon)$.

7.3 COALITIONS OF ARBITRARY SIZE

The example below shows that stable allocations may not exist if coalitions of arbitrary size are permitted. Agents in a coalition have to make an aggregate contribution of 1.

EXAMPLE 12 Let $N = \{1, 2, 3, 4\}$. Agents' preferences are symmetric. Table 12 summarizes the peaks of the agents.

In any stable allocation, agent 4 must have a contribution of 1. Otherwise the allocation will be blocked by the singleton coalition $\{4\}$ where she contributes 1.

p_1	p_2	p_3	p_4
0.55	0.55	0.55	1

Table 12: Peaks of agents in Example 12.

Consider an arbitrary allocation σ . We have just shown that $t_4 = 1$ if σ is stable. There are two cases to consider. The first case is where agent 4 belongs to a coalition C with some other agents in σ . All agents in $C \setminus \{4\}$ will have a contribution of 0. If |C| = 4 or |C| = 3, then any two agents from $C \setminus \{4\}$ will block with the contribution vector (0.5, 0.5). Assume w.l.o.g. $C = \{1, 4\}$. One of the agents in the set $\{2, 3\}$ (say 2) does not get her peak. The pair (1, 2) blocks with (0.45, 0.55).

The second case is where agent 4 is on her own in σ . If 1, 2 and 3 belong to the same coalition, there exists an agent $i \in \{1, 2, 3\}$ with $t_i \leq \frac{1}{3}$. Also there is at most one agent who receives her peak, i.e there exists $j \neq i$ with $t_j \neq 0.55$. The pair (i, j) blocks with (0.45, 0.55). In all other remaining cases, there exists an agent i who is on her own (her contribution is 1) and another agent j who does not get her peak. Then (i, j) can block with (0.45, 0.55). \Box

Our negative result in the case of arbitrary coalitions bears a resemblance to some earlier results on stability in division problems. Gensemer et al. (1996) consider the problem of allocating agents with single-peaked preferences across a set of islands. Each island has a unit amount of resource and operates with a fixed division rule. It can also accommodate an arbitrary number of agents. The paper formulates a notion of a migration equilibrium according to which no agent can benefit by migrating to another island. No island has the right to refuse an entrant. The paper provides a number of negative results about the existence of migration equilibria.

Bergantiños et al. (2015) consider a related model where all islands use the same division rule. They also weaken the equilibrium condition to a stability notion - in order for successful blocking to take place, the migrant's well-being must strictly improve while no member of the receiving island is made strictly worse-off by the move. The paper shows that stable allocations exist for some special division rules such as the proportional rule and the sequential dictatorship rule, provided agents' preferences are symmetric.

7.4 Strategy-proofness

An allocation rule is strategy-proof if no agent can strictly improve by misrepresenting her preferences. This property ensures that the mechanism designer can achieve the allocation specified at a preference profile by relying on the reports of the agents themselves. We show in Example 13 below that the SAM algorithm is not strategy-proof.¹⁹

We begin with an important observation regarding the SAM algorithm. It first selects a primary agent, then determines her contribution and finally chooses her partner. The contribution of the primary agent *i* depends whether she is a high or a low type agent: namely if $i \in H$ she gets $t_i = \max\{p_i, 1 - \inf U_q\}$ and is matched with the unmatched low type agent *j* with the largest $e_j(1-t_i)$; if $i \in L$ she gets $t_i = \min\{p_i, \inf D_q\}$ and is matched with the unmatched high type agent *j* with the largest $e_j(t_i)$. Suppose agents (i, j) are matched in Step *q*. Suppose also that $\inf U_q \leq \inf D_q$ and agent *i* is l_q . By letting $p_a = \inf U_q$ and $p_b = \inf D_q$, we observe that the allocation to agent l_q is exactly as specified in Equation 1, i.e. once a matching is formed, the SAM algorithm specifies the contributions of the matched pair according to one of the rules described in Barberà et al. (1997). However, the two numbers p_a, p_b depends on the allocation and the preferences of the agents that are already matched by the algorithm. This observation suggests that an agent's preference misreport can be profitable only if it induces a change in her matching. The following example shows that this can indeed happen in the SAM algorithm - by misreporting her preferences, an agent can be matched to a different partner and obtain a higher utility.

EXAMPLE 13 Let $N = \{1, 2, 3, 4\}$. Agents' preferences \succeq_i are symmetric and Table 13 summarizes their peaks. The ordering of the agents is $1 \succ^N 2 \succ^N 3 \succ^N 4$.

p_1	p_2	p_3	p_4
0.59	0.58	0.57	0.4

Table 13: Peaks of agents in Example 13.

In Step 0, agents 2 and 3 are removed from H and paired together. This is because $e_3(0.5) > e_2(0.5) > e_1(0.5)$. The triple (2, 3, 0.5) is formed and $U_1 = (0.34, 0.5)$ and $D_1 = \emptyset$. Since $p_4 = 0.4 > \inf U_1 = 0.34$, Substep 1.2 applies. Agent $4 \in L$ is the primary agent and $t_4 = \min\{p_4, \inf D_1\} = 0.4$. The triple (4, 1, 0.4) is formed. The allocation generated by the SAM algorithm is (2, 3, 0.5), (4, 1, 0.4).

Suppose agent 2 reports a peak of 0.6 and symmetric preferences \succeq'_2 . Now in Step 0, the triple (1, 3, 0.5) is formed and $U_1 = (0.32, 0.5)$. Substep 1.2 applies as $p_4 > \inf U_1 = 0.32$. The triple (2, 4, 0.4) is formed. Agent 2 strictly improves at \succeq_2 by misreporting since $0.6 \succ_2 0.5$. \Box

The existence of a strategy-proof, stable and Pareto-efficient rule in our model remains an open question. However, it is easy to show the existence of rules that are Pareto-efficient

¹⁹According to the definition, the algorithm specifies an allocation at a preference profile. We are slightly abusing terms here by regarding the algorithm as an allocation rule.

and strategy-proof. We describe such a rule in the case of four agents. It can be extended to an arbitrary even number of agents but we omit the details since they are cumbersome.

- 1. Agent 1 is always assigned her peak.
- 2. If the sum of the peaks of 1 and 2 is one, they are matched together and contribute their respective peaks. Agents 3 is matched with 4 and 3's contribution is her peak.
- 3. If case 2 above does not occur, 3 is matched with either 1 or 2, and her contribution is the complement of her partner's peak. Specifically, 3 is assigned to the triple she strictly prefers between $(1, 3, p_1)$ and $(2, 3, p_2)$.
- 4. In case 3 above, if 3 is indifferent between $(1, 3, p_1)$ and $(2, 3, p_2)$, then 4 chooses between the triples $(1, 4, p_1)$ and $(2, 4, p_2)$. Agent 3 is matched with the remaining agent and her contribution is still equal to the complement of her partner's peak.
- 5. If both 3 and 4 are both indifferent in cases 3 and 4, a tie-breaking rule determines the agent to who they are matched.

We claim that the rule is strategy-proof. Note that 1 and 2 always receive their peak so they cannot gain by misreporting. Agent 3 chooses the preferred alternative between two alternatives that do not depend on her reporting. The same holds for 4 in case 3 is indifferent. The rule also satisfies Pareto-efficiency holds because the rule is a sequential dictatorship that takes into account the potential efficiency losses due to mismatches (when it is efficient to match together agents 1 and 2) or due to indifferences.

The rule does not however, generate stable allocations. Consider the case where $p_1 = 0.9, p_2 = 0.9, p_3 = 0.49, p_4 = 0.49$. We can assume w.l.o.g that the allocation is (1, 3, 0.9) and (2, 4, 0.9). Then 3 and 4 will block with the triple (3, 4, 0.5).

7.5 Non-convexity of the set of stable allocations

The next example shows that a convex combination of two stable allocations with the same set of matched pairs may not be stable.

EXAMPLE 14 Let $N = \{1, 2, 3, 4\}$. The peaks of the agents are summarized in Table 14. All agents have symmetric preferences.

It is easy to verify that the allocations $\sigma^1 = \{(1, 4, 0.3), (2, 3, 0.3)\}$ and $\sigma^2 = \{(1, 4, 0.2), (2, 3, 0.1)\}$ are both stable. However the allocation $\sigma^3 = \{(1, 4, 0.25), (2, 3, 0.2)\}$ is not stable because the pair (2, 4) can block with (0.21, 0.79). Note that σ^3 has the same matched pairs as σ^1 and σ^2 but the contribution vector of each pair is a convex combination of the respective contributions in σ^1 and σ^2 with weights (0.5, 0.5).

p_1	p_2	p_3	p_4
0.3	0.3	0.8	0.9

Table 14: Peaks of agents in Example 14.

8 Appendix A

In this section, we provide a proof of the stability part of Theorem 1, i.e. the SAM algorithm generates a stable allocation. We begin with a few key observations.

Recall that $D_{q+1} = D_q \cup I_{l_q,t_{l_q}}$ and $U_{q+1} = U_q \cup I_{h_q,t_{h_q}}$ for $q \in \{0, 1, \ldots, K\}$. For every step q of the algorithm where $q \in \{1, \ldots, K\}$, we have $D_q = \bigcup_{r < q} I_{l_r,t_{l_r}}$ and $U_q = \bigcup_{r < q} I_{h_r,t_{h_r}}$. Since the improvement sets are open (see Observation 1), it follows that D_q and U_q are also open in [0, 0.5].

The sets D_q and U_q can be written as the disjoint union of their connected components. Since D_q and U_q are open sets, none of their connected components are singletons - thus each connected component of D_q and U_q is an interval in [0,0.5]. Moreover the connected components of D_q and U_q can be ordered from "left" to "right". Let D_q^1 and U_q^1 denote the "leftmost" connected components of D_q and U_q respectively. By definition, $\inf D_q = \inf D_q^1$ and $\sup D_q^1 \leq \inf D_q^r$ for any component D_q^r other than D_q^1 . Similar inequalities hold for U_q^1 . Incase D_q or U_q is empty (then D_q^1 or U_q^1 do not exist), we adopt the convention that the infimum and supremum of D_q^1 and U_q^1 is $+\infty$.

OBSERVATION 2 Consider step q where $q \in \{1, \ldots, K\}$. Recall that the triple formed in this step is (l_q, h_q, t_{l_q}) . If $p_{l_q} \ge \inf D_q^1$ then $\sup D_q^1 \le \sup D_{q+1}^1$. Similarly, if $1 - p_{h_q} \ge \inf U_q^1$ then $\sup U_q^1 \le \sup U_{q+1}^1$. This is an immediate consequence of the definition of improvement sets.

We now establish a series of results that are loop invariants of the algorithm.

LEMMA 1 Fix $q \in \{1, \ldots, K\}$ and assume $D_q \cap U_q = \emptyset$. Then for all $q \in \{1, \ldots, K\}$, we have

$$[\forall h \in H_q, \ 1 - p_h < \sup U_q^1] \ and \ [\forall l \in L_q, \ p_l < \sup D_q^1].$$

Proof: We will prove the lemma by induction on q.

BASE CASE (q = 1): There are two cases to consider $-U_1 = \emptyset$ and $U_1 \neq \emptyset$. If the former holds, then $[\forall h \in H_1, 1 - p_h < \sup U_1^1]$ is true since $\sup U_1^1 = +\infty$. Suppose $U_1 \neq \emptyset$. All agents allocated in Step 0 have a contribution of 0.5. Hence $\sup U_1^1 = 0.5$ and $1 - p_h \leq 0.5 = \sup U_1^1$ for all $h \in H$. Suppose there exists an agent $h' \in H_1$ and $1 - p_{h'} = 0.5$, i.e. $e(p_{h'}) = 0.5$. Since $U_1 \neq \emptyset$, there exists an agent \bar{h} allocated in Step 0 for whom $p_{\bar{h}} > 0.5$, i.e. $e_{\bar{h}}(0.5) < 0.5$. But then h' has higher priority than \bar{h} in H and should

have been allocated in Step 0. Therefore $[\forall h \in H_1, 1 - p_h < \sup U_1^1]$ holds. The argument for L_1 is identical and omitted.

INDUCTIVE STEP: Consider $q \in \{1, \ldots, K\}$. Assume $D_q \cap U_q = \emptyset$, $[\forall h \in H_q, 1 - p_h < \sup U_a^1]$ and $[\forall l \in L_q, p_l < \sup D_a^1]$. We have to show

$$[\forall h \in H_{q+1}, 1 - p_h < \sup U_{q+1}^1] \text{ and } [\forall l \in L_{q+1}, p_l < \sup D_{q+1}^1].$$

We refer to $[\forall h \in H_{q+1}, 1-p_h < \sup U_{q+1}^1]$ and $[\forall l \in L_{q+1}, p_l < \sup D_{q+1}^1]$ as Statements A and B respectively. There are two cases to consider depending on whether D_q^1 lies to the left or to the right of U_q^1 .

Case I: $\sup D_q^1 \leq \inf U_q^1$ (see Figure 3). There are two sub-cases to consider depending on the selection of the primary agent.



Figure 3: Case I in the proof of Lemma 1

Case I.a: The primary agent is h_q . If $1-p_{h_q} > \inf U_q^1$, Substep q.1 applies and the contribution of h_q is $1 - \inf U_q^1$. By Observation 2, $\sup U_q^1 \le \sup U_{q+1}^1$. Statement A follows from the induction hypothesis and $H_{q+1} \subset H_q$. If $1 - p_{h_q} \le \inf U_q^1$, then the contribution of h_q is p_{h_q} . Consequently the improvement set of h_q is empty and $\sup U_q^1 = \sup U_{q+1}^1$. Statement A once again follows from the induction hypothesis and $H_{q+1} \subset H_q$.

We now prove Statement *B* for the secondary agent l_q . By the induction hypothesis, $p_{l_q} < \sup D_q^1$. If $p_{l_q} \ge \inf D_q^1$, then $p_{l_q} \in D_q^1$. Observation 2 implies $\sup D_q^1 \le \sup D_{q+1}^1$ and Statement *B* holds following the earlier argument. Suppose $p_{l_q} < \inf D_q$. There are two possibilities depending on the location of the peak of h_q .

(i) If $1 - p_{h_q} > \inf D_q^1$, then Substep q.1 is applicable. Here $t_{l_q} \ge \inf D_q^1$ as $t_{l_q} = \min\{1 - p_{h_q}, \inf U_q\}$. This implies $\sup D_q^1 \le \sup D_{q+1}^1$ and Statement B follows from the induction hypothesis.

(ii) The remaining case is when $1 - p_{h_q} \leq \inf D_q^1$. This can happen only if Substep q.3.1 occurs. In particular, we have $p_l \leq 1 - p_{h_q}$ for all $l \in L_q$. So $p_{l_q} \leq 1 - p_{h_q}$. The contribution of agent l_q is $t_{l_q} = 1 - p_{h_q}$. If $p_{l_q} = 1 - p_{h_q}$, then the improvement set of l_q is empty and Statement *B* follows from the induction hypothesis and the fact that $D_{q+1}^1 = D_q^1$. Suppose $p_{l_q} < 1 - p_{h_q} = t_{l_q}$. Then $\sup D_{q+1}^1 = t_{l_q}$ and $p_l \leq \sup D_{q+1}^1$ for all $l \in L_q$. Suppose there exists $\bar{l} \in L_q$ with $p_{\bar{l}} = \sup D_{q+1}^1 = t_{l_q}$. Then $e_{\bar{l}}(t_{l_q}) = t_{l_q}$. However $e_{l_q}(t_{l_q}) < t_{l_q}$ as $p_{l_q} < t_{l_q}$.

Then l should have been matched with h_q in Step q instead of l_q . This establishes Statement B in this case.

Case I.b: The primary agent is l_q . This case is symmetric to the case when h_q is the primary agent. If $p_{l_q} \ge \inf D_q^1$, then the result follows from Observation 2. If $p_{l_q} < \inf D_q^1$, then l_q receives her peak and the improvement set is empty. Statement *B* again follows immediately.

Here h_q is the secondary agent. This can happen only if Step q.3.2 occurs. In particular, $1 - p_h < p_{l_q}$ and $1 - p_h \leq \inf D_q$ for all $h \in H_q$. Note that $t_{l_q} = \min\{p_{l_q}, \inf D_q\}$. Thus $1 - p_h \leq t_{l_q}$ for all $h \in H_q$ and $\sup U_{q+1}^1 = t_{l_q}$. Using arguments like those in Case I.a.(ii) above, we can show $1 - p_h < t_{l_q}$ for all $h \in H_q$. This establishes Statement A.

Case II: $\sup U_q^1 \leq \inf D_q^1$ (see Figure 4). Once again there are two cases depending upon the selection of the primary agent.



Figure 4: Case II in the proof of Lemma 1

The arguments for the primary agent are symmetric to those for the primary agent in Case I. This is also true for the secondary agent. For instance, consider the case where l_q is the secondary agent. This occurs only if Step q.3.1 occurs. In particular, $p_l \leq 1 - p_{h_q}$ and $p_l \leq \inf U_q$ for all $l \in L_q$. So $p_l \leq t_{l_q}$ for all $l \in L_q$. Hence $\sup D_{q+1}^1 = t_{l_q}$. Using arguments similar to those in Case I.b.(ii), we can verify Statement B.

We omit the other arguments.

LEMMA 2 Consider Step q where $q \in \{1, ..., K\}$. Assume $D_q \cap U_q = \emptyset$. Then the triple formed in Step q satisfies the following:

1. if it is formed in Substep q.1, then either $p_{l_q} \leq t_{l_q} \leq 1 - p_{h_q}$ or $t_{l_q} = 1 - p_{h_q} < p_{l_q}$.

2. if it is formed in Substep q.2, then either $1 - p_{h_q} \leq t_{l_q} \leq p_{l_q}$ or $t_{l_q} = p_{l_q} < 1 - p_{h_q}$.

3. if it is formed in Substep q.3.1, then $p_{l_q} \leq t_{l_q} \leq 1 - p_{h_q}$.

4. if it is formed in Substep q.3.2, then $1 - p_{h_q} \leq t_{l_q} \leq p_{l_q}$.

Proof: Since $D_q \cap U_q = \emptyset$, it must be the case that D_q^1 either lies entirely to the "left" of U_q^1 or entirely to the "right" of U_q^1 . We now consider each of the four cases in turn.

1. Suppose the triple (l_q, h_q, t_{l_q}) is formed in Substep q.1. Lemma 1 rules out the case where D_q^1 lies entirely to the right of U_q^1 .

There are now two possibilities. The first is $1 - p_{h_q} < p_{l_q}$. By Lemma 1, we know $p_{l_q} < \sup D_q^1$. Since $t_{l_q} = \min\{1 - p_{h_q}, \inf U_q\}$ and $\sup D_q^1 \leq \inf U_q$, we have $t_{l_q} = 1 - p_{h_q}$. Thus $t_{l_q} = 1 - p_{h_q} < p_{l_q}$.

The second possibility is $1 - p_{h_q} \ge p_{l_q}$. If $1 - p_{h_q} \ge \inf U_q$, we have $t_{l_q} = \inf U_q$. By Lemma 1, $p_{l_q} < \sup D_q^1$. Thus $p_{l_q} \le t_{l_q} \le 1 - p_{h_q}$. If $1 - p_{h_q} < \inf U_q$, we have $t_{l_q} = 1 - p_{h_q}$ and once again $p_{l_q} \le t_{l_q} = 1 - p_{h_q}$.

2. Suppose the triple (l_q, h_q, t_{l_q}) is formed in Substep q.2. In this case, Lemma 1 implies D_q^1 must lie to the right of U_q^1 . We can use the symmetric counterparts of the arguments in Case 1 to derive the necessary conclusion.

3. Suppose the triple (l_q, h_q, t_{l_q}) is formed in Substep q.3.1. Here $t_{l_q} = \min\{\inf U_q, 1-p_{h_q}\}$. By the hypothesis of Substep q.3.1, we know $p_l \leq \inf U_q$ for all $l \in L_q$. Also $p_l \leq 1-p_{\tilde{h}_q} = 1-p_{h_q}$ for all $l \in L_q$.²⁰ In particular, $p_{l_q} \leq \inf U_q$ and $p_{l_q} \leq 1-p_{h_q}$. Thus $p_{l_q} \leq \min\{\inf U_q, 1-p_{h_q}\} = t_{l_q} \leq 1-p_{h_q}$.

4. Suppose the triple (l_q, h_q, t_{l_q}) is formed in Substep q.3.2. Here $t_{l_q} = \min\{\inf D_q, p_{l_q}\}$. We can use the symmetric counterparts of the arguments in Case 3 to derive the necessary conclusion.

Lemma 2 immediately leads to the following corollary.

COROLLARY 1 Consider Step q where $q \in \{1, \ldots, K\}$. Assume $D_q \cap U_q = \emptyset$. Then the contribution of agent l_q , t_{l_q} lies in the closed interval with the end points p_{l_q} and $1 - p_{h_q}$.

LEMMA **3** Consider Step q where $q \in \{1, ..., K\}$. Assume $D_q \cap U_q = \emptyset$. 1. If $t_{l_q} \leq p_{l_q}$, then $I_{l_q, t_{l_q}} \subseteq D_q$. 2. If $t_{l_q} \leq 1 - p_{h_q}$, then $I_{h_q, t_{h_q}} \subseteq U_q$.

Proof: We only consider Part 1 - a symmetric argument applies for Part 2. Consider Step q where the triple (l_q, h_q, t_{l_q}) is formed.

If $p_{l_q} = t_{l_q}$, the improvement set of l_q is empty and the result follows immediately. Assume therefore that $t_{l_q} < p_{l_q}$. By Corollary 1, we have $1 - p_{h_q} \le t_{l_q} < p_{l_q}$.

We will argue that D_q is non-empty when $t_{l_q} < p_{l_q}$. By Lemma 2, we know Substep q.3.1 cannot occur as $t_{l_q} < p_{l_q}$. We establish the claim for Substeps q.1, q.2 and q.3.2. If Step q.1 occurs, we know $1 - p_{h_q} > \inf D_q$. Thus $\inf D_q$ is not $+\infty$ and D_q is non-empty. If Step q.2 or Step q.3.2 occurs, we know $t_{l_q} = \min\{p_{l_q}, \inf D_q\}$. Since $t_{l_q} < p_{l_q}$, we have $t_{l_q} = \inf D_q$ and thus D_q is non-empty.

Claim 1: If $t_{l_q} < p_{l_q}$, then $t_{l_q} \ge \inf D_q$.

²⁰Recall \tilde{h}_q is the high type agent with the lowest peak in h_q and $h_q = \tilde{h}_q$ in Substep q.3.1.

Proof: Lemma 2 implies that the triple cannot be formed in Substep q.3.1. We establish the claim for Substeps q.1, q.2 and q.3.2.

Suppose the triple is formed in Substep q.1. Since $D_q \cap U_q = \emptyset$ by assumption, Lemma 1 implies D_q^1 lies entirely to the left of U_q^1 and $\sup D_q^1 \leq \inf U_q^1$. In Step q.1, we know $1 - p_{h_q} > \inf D_q$ and $t_{l_q} = \min\{1 - p_{h_q}, \inf U_q\}$. Thus $t_{l_q} > \inf D_q$.

Suppose the triple is formed in Substep q.2. Here $t_{l_q} = \min\{p_{l_q}, \inf D_q\}$. Since $t_{l_q} < p_{l_q}$ by assumption, it must be the case that $t_{l_q} = \inf D_q$.

Suppose the triple is formed in Substep q.3.2. Here $t_{l_q} = \min\{p_{l_q}, \inf D_q\}$. Since $t_{l_q} < p_{l_q}$ by assumption, it follows that $t_{l_q} = \inf D_q$. This completes the proof of the claim.

Since D_q is a finite union of intervals, $\inf D_q$ is the infimum of at least one of these intervals. Thus $\inf D_q$ is attained in a step before q. Let i be the smallest integer such that $\inf I_{l_i,t_{l_i}} = \inf D_q$ where l_i is a low type agent matched in Step i.²¹ Note that $\inf D_j > \inf D_q$ for $j \in \{1, \ldots, i\}$ and $\inf D_{i+1} = \inf D_q$ when $i \ge 1$. If i = 0, then $\inf D_1 = \inf D_q$.

Consider Step *i* where agent l_i is matched and her contribution is t_{l_i} . We claim that $t_{l_i} > p_{l_i}$. Suppose not. If $t_{l_i} = p_{l_i}$, the improvement set of l_i is empty, contradicting the assumption that $\inf I_{l_i,t_{l_i}} = \inf D_q$. Suppose $t_{l_i} < p_{l_i}$. Here $\inf I_{l_i,t_{l_i}} = t_{l_i}$. Arguing as we did to establish Claim 1, it follows $t_{l_i} \ge \inf D_i$ in Step *i*. Therefore $t_{l_i} = \inf D_q \ge \inf D_i$. This leads to a contradiction since $\inf D_i > \inf D_q$.

Since $t_{l_i} > p_{l_i}$, we have $e_{l_i}(t_{l_i}) \leq \inf I_{l_i, t_{l_i}} = \inf D_q$.²²

Claim 2: For agents l_i and l_q , we have (i) $e_{l_q}(t_{l_i}) \leq e_{l_i}(t_{l_i}) \leq \inf D_q$ and (ii) $p_{l_q} < t_{l_i}$.

Proof: Recall that agent l_i is matched in Step *i*.

Suppose i = 0. Then it must be the case that |L| > |H| and $D_1 \neq \emptyset$. Every agent matched in this step has a contribution of 0.5. There exists $l_0 \in \overline{L}$ such that $\inf I_{l_0,0.5} = \inf D_1$. Since the improvement set of l_0 is non-empty, $p_{l_0} < 0.5$ and $e_{l_0}(0.5) < 0.5$. As l_q is not matched in Step 0, we have $e_{l_q}(0.5) \le e_{l_0}(0.5) < 0.5$. Hence $p_{l_q} < 0.5$. This establishes Claim 2 for i = 0.

Suppose $i \ge 1$. Recall that $t_{l_i} > p_{l_i}$. By Lemma 2, we know that l_i is matched in Substep i.1 or Substep i.3.1. In both cases, l_i is the secondary agent. Also $l_q \in L_i$ as i < q. Thus $e_{l_q}(t_{l_i}) \le e_{l_i}(t_{l_i})$. This establishes Part (i).

Now consider Part (ii). If $p_{l_q} = t_{l_i}$, then $e_{l_q}(t_{l_i}) = t_{l_i}$. Since $p_{l_i} < t_{l_i}$, then $e_{l_i}(t_{l_i}) < t_{l_i}$ and l_i should not have been chosen as the secondary agent in Step *i*. If $p_{l_q} > t_{l_i}$, we know $e_{l_q}(t_{l_i}) > t_{l_i}$. Once again l_i should not have been chosen as the secondary agent in Step *i*. This establishes Part (ii).

We now return to the proof of the lemma. Claims 1, 2 and the assumption $t_{l_q} < p_{l_q}$ imply

²¹If i = 0, note that l_i may not be unique. If $i \ge 1$, then l_i is unique.

²²Note that $e_{l_i}(t_{l_i}) < \inf I_{l_i, t_{l_i}}$ if and only if $e_{l_i}(t_{l_i}) = -\epsilon$. Here $I_{l_i, t_{l_i}} = [0, t_{l_i})$.

$$e_{l_q}(t_{l_i}) \le e_{l_i}(t_{l_i}) \le \inf D_q \le t_{l_q} < p_{l_q} < t_{l_i}.$$

Since \succeq_{l_q} is single-peaked, we have $t_{l_q} \succeq_{l_q} \max\{0, e_{l_q}(t_{l_i})\}$. Also $\max\{0, e_{l_q}(t_{l_i})\} \succeq_{l_q} t_{l_i}$.²³ Thus $t_{l_q} \succeq_{l_q} t_{l_i}$ and $t_{l_i} \notin I_{l_q,t_{l_q}}$. Consequently $\sup I_{l_q,t_{l_q}} \leq t_{l_i} = \sup I_{l_i,t_{l_i}}$. We have already shown $\inf I_{l_i,t_{l_i}} = \inf D_q \leq t_{l_q} = \inf I_{l_q,t_{l_q}}$. Therefore $I_{l_q,t_{l_q}} \subseteq I_{l_i,t_{l_i}} \subseteq D_q$.

We complete the proof of the theorem by showing that the SAM algorithm generates an allocation which satisfies Condition S (Proposition 3).

Proof: Recall that the algorithm terminates in K steps, generating the sets D_{K+1} and U_{K+1} . We will show that $D_{K+1} \cap U_{K+1} = \emptyset$ and $0.5 \notin D_{K+1} \cup U_{K+1}$. This guarantees that Condition S is satisfied by the allocation generated.²⁴

In fact, we will show that $D_q \cap U_q = \emptyset$ and $0.5 \notin D_q \cup U_q$ holds for all $q \in \{1, \ldots, K+1\}$. We will use induction on q for this purpose.

We claim that $D_1 \cap U_1 = \emptyset$ and $0.5 \notin D_1 \cup U_1$. By construction, at least one of the sets D_1 and U_1 is empty. Also the contribution of any agent matched in Step 0 is 0.5. Thus the improvement sets of agents matched in this step do not contain 0.5.

Induction Hypothesis (IH): Fix $q \in \{1, \ldots, K\}$ and assume (i) $D_q \cap U_q = \emptyset$ and (ii) $0.5 \notin D_q \cup U_q$.

The IH implies that the antecedents of Lemmas 1, 2, 3 and Corollary 1 are satisfied. We show that $D_{q+1} \cap U_{q+1} = \emptyset$ and $0.5 \notin D_{q+1} \cup U_{q+1}$. We have,

$$\begin{aligned} D_{q+1} \cap U_{q+1} &= (D_q \cup I_{l_q, t_{l_q}}) \cap (U_q \cup I_{h_q, t_{h_q}}) \\ &= (D_q \cap U_q) \cup (D_q \cap I_{h_q, t_{h_q}}) \cup (I_{l_q, t_{l_q}} \cap U_q) \cup (I_{l_q, t_{l_q}} \cap I_{h_q, t_{h_q}}) \end{aligned}$$

From IH, it follows $D_q \cap U_q = \emptyset$. By Corollary 1 and the fact that the improvement sets are open intervals, we have $I_{l_q,t_{l_q}} \cap I_{h_q,t_{h_q}} = \emptyset$. We will show (A) $I_{l_q,t_{l_q}} \cap U_q = \emptyset$, $0.5 \notin I_{l_q,t_{l_q}}$ and (B) $I_{h_q,t_{h_q}} \cap D_q = \emptyset$, $0.5 \notin I_{h_q,t_{h_q}}$.

We first prove (A). There are two cases to consider depending on the contribution of agent l_q . The first case is when $t_{l_q} \leq p_{l_q}$. Part 1 of Lemma 3 implies $I_{l_q,t_{l_q}} \subseteq D_q$. Thus $I_{l_q,t_{l_q}} \cap U_q = \emptyset$ and $0.5 \notin I_{l_q,t_{l_q}}$ since $D_q \cap U_q = \emptyset$ and $0.5 \notin D_q$ by IH.

In the second case, $p_{l_q} < t_{l_q}$. Here $t_{l_q} = \sup I_{l_q,t_{l_q}}$. Lemma 2 implies only Substeps q.1 and q.3.1 can occur. In both steps, we have $t_{l_q} = \min\{1 - p_{h_q}, \inf U_q\}$. Since $1 - p_{h_q} \le 0.5$, it follows that $t_{l_q} \le \min\{\inf U_q, 0.5\}$. Furthermore, $t_{l_q} = \sup I_{l_q,t_{l_q}}$ and $I_{l_q,t_{l_q}}$ is an open set. Therefore $x < t_{l_q} \le \min\{\inf U_q, 0.5\}$ for all $x \in I_{l_q,t_{l_q}}$. Thus $I_{l_q,t_{l_q}} \cap U_q = \emptyset$ and $0.5 \notin I_{l_q,t_{l_q}}$.

²³If max $\{0, e_{l_q}(t_{l_i})\} = 0$, then $e_{l_q}(t_{l_i}) = -\epsilon$ and $I_{l_i, t_{l_i}} = [0, t_{l_i})$. Here $0 \succ_{l_q} t_{l_i}$.

²⁴Recall D_{K+1} and U_{K+1} are the unions of the improvement sets of all L and H type agents respectively.

The proof of (B) is virtually identical to the arguments for (A), but uses Part 2 of Lemma 2. We omit the details. This completes the proof of the theorem. ■

We have shown above that the sets D_q and U_q do not intersect for any $q \in \{1, \ldots, K\}$. Thus the antecedents of Lemmas 1, 2, 3 and Corollary 1 are true. We will use these facts in the proof of Pareto efficiency in Appendix B.

9 Appendix B

This section contains the proof of the Pareto efficiency part of Theorem 1.

We define some notation which will be used in the proof of Pareto efficiency. For any allocation σ and agent $i \in N$, we shall denote the contribution of agent i in σ by t_i^{σ} . The improvement set for agent i at t_i^{σ} is $I_{i,t_i^{\sigma}}$ and it's closure is $\overline{I_{i,t_i^{\sigma}}}$. Note that $\overline{I_{i,t_i^{\sigma}}} = \{x \in [0, 0.5] : x \succeq_i t_i^{\sigma}\}$ when $i \in L$ and $\overline{I_{i,t_i^{\sigma}}} = \{x \in [0, 0.5] : 1 - x \succeq_i t_i^{\sigma}\}$ when $i \in H$. For the allocation σ , we define sets $D^{\sigma} = \bigcup_{i \in L} I_{l,t_i^{\sigma}}$ and $U^{\sigma} = \bigcup_{i \in H} I_{l,t_h^{\sigma}}$.

We first establish several key observations and lemmas.

OBSERVATION **3** In any stable allocation, agents who are paired together must be given contributions on the same side of the peak. We refer to this property as *internal stability* for the pair of agents who are matched together in the allocation. Internal stability is a necessary condition for both stability and Pareto efficiency.

Our next step is to establish a monotonicity lemma.

LEMMA 4 For any Step q where $q \in \{1, ..., K-1\}$, we have (i) $t_{l_{q+1}} \leq t_{l_q} \leq 0.5$ and (ii) $0.5 \leq t_{h_q} \leq t_{h_{q+1}}$.

Proof: We first prove Part (i). Suppose q = 1. The allocation to agent l_1 in Step 1 is either min{inf D_1, p_{l_1} } or min{inf $U_1, 1 - p_{h_1}$ }. Both these values are smaller or equal to 0.5.

Suppose $q \ge 1$. The triple (l_q, h_q, t_{l_q}) is formed in Step q. Next we show that $t_{l_q+1} \le t_{l_q}$. We have to consider four cases based on which substep occurs in Step q.

Case 1: Suppose the allocation is made in Substep q.1.

Here $t_{l_q} = \min\{\inf U_q, 1 - p_{h_q}\} \le 1 - p_{h_q}$. By Lemma 3, we know $U_{q+1} = U_q$ and $\inf U_{q+1} = \inf U_q$.

Lemma 1 and the hypothesis of Substep q.1 imply $\inf D_q < 1 - p_{h_q} < \sup U_q^1$. Since $D_q \cap U_q = \emptyset$, it must be the case that D_q^1 lies to the left of U_q^1 , i.e. $\inf D_q < \inf U_q$. Thus $\inf D_q < \min\{\inf U_q, 1 - p_{h_q}\} = t_{l_q}$.

Now we consider the triple $(l_{q+1}, h_{q+1}, t_{l_{q+1}})$ formed in Step q + 1. There are four possibilities to consider.

- 1. The allocation is made in Substep q + 1.1. By hypothesis of the substep, we have $\inf D_{q+1} < 1 - p_{h_{q+1}}$. If $1 - p_{h_{q+1}} \leq \inf D_q$, then $1 - p_{h_q} < \inf U_q = \inf U_{q+1}$ and $t_{l_{q+1}} = 1 - p_{h_{q+1}}$. Thus $t_{l_{q+1}} = 1 - p_{h_{q+1}} \leq \inf D_q < t_{l_q}$. If $1 - p_{h_{q+1}} > \inf D_q$, then $h_{q+1} \in \{h \in H_q : 1 - p_h > \inf D_q\}$. Since h_q is matched in Step q, it must be the case that $1 - p_{h_{q+1}} \leq 1 - p_{h_q}$. Recall $\inf U_{q+1} = \inf U_q$. Thus $t_{l_{q+1}} = \min\{\inf U_{q+1}, 1 - p_{h_{q+1}}\} \leq \min\{\inf U_q, 1 - p_{h_q}\} = t_{l_q}$.
- 2. The allocation is made in Substep q + 1.2. Note than in Step q, D_q^1 lies to the left of U_q^1 . Since $D_q \cap U_q = \emptyset$, we have $\sup D_q^1 \leq \inf U_q^1$. Thus $p_l < \sup D_q^1 \leq \inf U_q^1$ for all $l \in L_q$ (Lemma 1). Since $U_{q+1} = U_q$ and $L_{q+1} \subset L_q$, there does not exist $l \in L_{q+1}$ such that $p_l > \inf U_{q+1}$. So Substep q + 1.2 is not possible.
- 3. The allocation is made in Substep q+1.3.1. By the hypothesis of the substep, we know $1-p_{h_{q+1}} \leq \inf D_{q+1}$. Note that $\inf D_{q+1} \leq \inf D_q$.²⁵ Thus

$$t_{l_q+1} = \min\{\inf U_{q+1}, 1 - p_{h_{q+1}}\} \le \inf D_{q+1} \le \inf D_q < t_{l_q}$$

4. The allocation is made in Substep q + 1.3.2. Here $t_{l_{q+1}} = \min\{\inf D_{q+1}, p_{l_{q+1}}\}$. Also $\inf D_{q+1} \leq \inf D_q$. Thus

$$t_{l_{q+1}} \le \inf D_{q+1} \le \inf D_q < t_{l_q}.$$

Case 2: The allocation is made in Substep q.3.1.

Here $t_{l_q} = \min\{\inf U_q, 1 - p_{h_q}\} \le 1 - p_{h_q}$. By Lemma 3, we have $U_{q+1} = U_q$ and $\inf U_{q+1} = \inf U_q$.

Consider the allocation is Step q + 1. There are four possibilities.

1. The allocation is done in Substep q + 1.1. Here $t_{l_{q+1}} = \min\{\inf U_{q+1}, 1 - p_{h_{q+1}}\} \le 1 - p_{h_{q+1}}$. Since agent h_q is matched in Step q and $h_{q+1} \in H_q$, it must be the case that $1 - p_{h_{q+1}} \le 1 - p_{h_q}$. Thus

$$t_{l_{q+1}} = \min\{\inf U_{q+1}, 1 - p_{h_{q+1}}\} \le \{\inf U_q, 1 - p_{h_q}\} = t_{l_q}.$$

2. The allocation is made in Subtep q + 1.2. We will show that this is not possible. Since the first allocation is done in Substep q.3.1, we know $p_l \leq \inf U_q$ for all $l \in L_q$. Note that $\inf U_{q+1} = \inf U_q$ and $L_{q+1} \subset L_q$. Thus there does not exist $l \in L_{q+1}$ such that $p_l > \inf U_{q+1}$.

²⁵Recall $D_q \subseteq D_{q+1}$. Thus $\inf D_{q+1} \leq D_q$.

3. The allocation is made in Substep q+1.3.1. Here $t_{l_{q+1}} = \min\{\inf U_{q+1}, 1-p_{h_{q+1}}\}$. Since $h_{q+1} \in H_q$ and agent h_q is matched in Substep q.3.1, we have $1-p_{h_{q+1}} \leq 1-p_{h_q}$. Thus

$$t_{l_{q+1}} = \min\{\inf U_{q+1}, 1 - p_{h_{q+1}}\} \le \{\inf U_q, 1 - p_{h_q}\} = t_{l_q}$$

4. The allocation is made in Substep q+1.3.2. Here $t_{l_{q+1}} = \min\{\inf D_{q+1}, p_{l_{q+1}}\}$. Since the first allocation was done in Substep q.3.1 and $l_{q+1} \in L_q$, we know $p_{l_{q+1}} \leq p_{\tilde{l}_q} \leq 1 - p_{h_q}$ and $p_{l_{q+1}} \leq \inf U_q$.²⁶ Thus

$$t_{l_{q+1}} = \min\{\inf D_{q+1}, p_{l_{q+1}}\} \le p_{l_{q+1}} \le \min\{\inf U_q, 1 - p_{h_q}\} = t_{l_q}.$$

Case 3 occurs when the allocation is made in Substep q.2. Case 4 occurs when the allocation is made in Substep q.3.2. These cases can be argued similarly by making the appropriate changes. We omit the details. This completes the proof of Part (i) of the lemma.

Note that $t_{h_q} = 1 - t_{l_q}$ for any $q \in \{1, \ldots, K - 1\}$. Thus Part (i) of the lemma implies Part (ii).

OBSERVATION 4 Consider an allocation σ and an $x \in [0,1]$. Suppose there exists $i \in N$ with $t_i^{\sigma} = x$. Then agent *i*'s partner in σ , say agent *j* has contribution $t_j^{\sigma} = 1 - x$. Thus $|\{i \in N : t_i^{\sigma} = x\}| = |\{j \in N : t_j^{\sigma} = 1 - x\}|.$

LEMMA 5 Consider any preference profile \succeq . Let σ and τ be stable allocations at the preference profile \succeq . For all $x \in [0, 0.5]$ such that σ and τ satisfy

- (1) $t_i^{\sigma} \succeq_i x$ and $t_i^{\sigma} \succeq_i 1 x$ for all $i \in N$ and
- (2) $t_i^{\tau} \succeq_i x \text{ and } t_i^{\tau} \succeq_i 1 x \text{ for all } i \in N$

we have,

- (a) $|\{i \in N : t_i^{\sigma} = x\}| = |\{i \in N : t_i^{\tau} = x\}|.$
- (b) Consider agents $i, j \in N$ such that $p_i < 0.5$ and $p_j > 0.5$. If 0.5 satisfies (1) and (2), then $t_i^{\sigma} = t_j^{\tau} = 0.5$ is not possible.

Proof: Consider an arbitrary preference profile \succeq . Let σ and τ be stable allocations at \succeq . Also consider an $x \in [0, 0.5]$ such that σ satisfies Condition (1) and τ satisfies Condition (2) in the antecedent of the lemma.

We partition the agents in N into three groups depending on where their peaks lie with respect to the points x and 1 - x.²⁷ Define,

²⁶If $p_{l_{q+1}} > \inf U_q$, then the allocation is Step q would be done in Substep q.2.

²⁷Note that it is possible that x = 1 - x = 0.5. Then S_x^2 is the set of agents whose peaks are exactly 0.5.

1. $S_x^1 = \{i \in N | p_i < x\}.$ 2. $S_x^2 = \{i \in N | x \le p_i \le 1 - x\}.$ 3. $S_x^3 = \{i \in N | p_i > 1 - x\}.$

We will prove several claims about the allocation σ .

Claim 1: For the allocation σ , we have

t^σ_i ≤ x for all i ∈ S¹_x.
 x ≤ t^σ_i ≤ 1 − x for all i ∈ S²_x.
 1 − x ≤ t^σ_i for all i ∈ S³_x.

Proof: We will prove the claim by contradiction. We first prove Part 1. Consider an agent $i \in S_x^1$ such that $t_i^{\sigma} > x$. Here $p_i < x < t_i^{\sigma}$ and single-peakedness implies $x \succ_i t_i^{\sigma}$. This contradicts the fact that σ satisfies Condition (1) in the antecedent of the lemma.

For Part 2, consider $i \in S_x^2$ such that $t_i^{\sigma} < x$ or $t_i^{\sigma} > 1 - x$. If $t_i^{\sigma} < x$, we have $t_i^{\sigma} < x \le p_i$. Thus $x \succ_i t_i^{\sigma}$ by single-peakedness and we have a contradiction. If $t_i^{\sigma} > 1 - x$, we have $p_i \le 1 - x < t_i^{\sigma}$ and $1 - x \succ_i t_i^{\sigma}$. Once again we have a contradiction.

Part 3 can be proved using similar arguments.

Claim 2: For the allocation σ , we have

(a) Consider $i \in S_x^1$. Let (i, j, t_i^{σ}) be the triple that agent *i* belongs to in σ . If $t_i^{\sigma} \neq x$, then $j \in S_x^3$.

(b) Consider agents $i \in S_x^1$, $j \in S_x^3$ such that $(i, j, t_i^{\sigma}) \in \sigma$. Then $t_i^{\sigma} < x$.

Proof: (a) Consider $i \in S_x^1$. By assumption, $t_i^{\sigma} \neq x$. By Part 1 of Claim 1, we have $t_i^{\sigma} < x$. Thus $t_j^{\sigma} = 1 - t_i^{\sigma} > 1 - x$. From Claim 1, we know that all agents in S_x^1 receive a contribution of contribution of at most x in σ (Part 1). Similarly, all agents in S_x^2 have a contribution of at most 1 - x (Part 2). Finally all agents in S_x^3 have a contribution of at least 1 - x in σ (Part 3). Thus $j \in S_x^3$.

(b) Consider a triple $(i, j, t_i^{\sigma}) \in \sigma$ where $i \in S_x^1$ and $j \in S_x^3$. By internal stability for the pair (i, j), we have

$$\min\{p_i, 1 - p_j\} \le t_i^{\sigma} \le \max\{p_i, 1 - p_j\}.$$

Since $\max\{p_i, 1-p_j\} < x$ (follows from the definition of S_x^1 and S_x^3), we have $t_i^{\sigma} < x$.

Claim 3: For the allocation σ , we have

(a) Consider $i \in S_x^3$. Let (i, j, t_i^{σ}) be the triple that agent *i* belongs to in σ . If $t_i^{\sigma} \neq 1 - x$, then $j \in S_x^1$.

(b) Consider agents $i \in S_x^3$, $j \in S_x^1$ such that $(i, j, t_i^{\sigma}) \in \sigma$. Then $t_i^{\sigma} > 1 - x$.

Proof: (a) Consider $i \in S_x^3$. By assumption, $t_i^{\sigma} \neq 1 - x$. By Part 3 of Claim 1, we have $t_i^{\sigma} > 1 - x$. Thus $t_j^{\sigma} = 1 - t_i^{\sigma} < x$. From Claim 1, all agents in S_x^2 must have a contribution of at least x in σ (Part 2). Also all agents in S_x^3 have a contribution of at least 1 - x in σ (Part 3). Finally all agents in S_x^1 have a contribution of at most x in σ (Part 1). Thus $j \in S_x^1$.

(b) This follows from Part (b) of Claim 2.

Claim 4: For the allocation σ , there does not exist a pair of agents $i \in S_x^1$, $j \in S_x^3$ such that $t_i^{\sigma} = x$ and $t_j^{\sigma} = 1 - x$.

Proof: Suppose not. Then there exist $i \in S_x^1$, $j \in S_x^3$ such that $t_i^{\sigma} = x$ and $t_j^{\sigma} = 1 - x$. By the definition of S_x^1 and S_x^3 , we know $\max\{p_i, 1 - p_j\} < x$. Thus there exists $\epsilon > 0$ such that $x > x - \epsilon \ge \max\{p_i, 1 - p_j\}$. The pair (i, j) blocks σ with the contribution vector $(x - \epsilon, 1 - x + \epsilon)$. This results in a contradiction as σ is stable by assumption.

Claim 5: For the allocation σ ,

1. For any $i \in S_x^1$ and $j \in N \setminus S_x^3$ such that $(i, j, t_i^{\sigma}) \in \sigma$, we have $t_i^{\sigma} = x$.

2. For any $i \in N \setminus S_x^1$ and $j \in S_x^3$ such that $(i, j, t_i^{\sigma}) \in \sigma$, we have $t_i^{\sigma} = x$ and $t_j^{\sigma} = 1 - x$.

Proof: (1) Consider agents $i \in S_x^1$ and $j \in N \setminus S_x^3$ such that $(i, j, t_i^{\sigma}) \in \sigma$. By Claim 1 and $i \in S_x^1$, we know $t_i^{\sigma} \leq x$. Agent j belongs to S_x^1 or S_x^2 . If $j \in S_x^2$, Claim 1 implies $x \leq t_j^{\sigma} \leq 1 - x$. These together imply $t_i^{\sigma} = x$. If $j \in S_x^1$, we have $t_j^{\sigma} \leq x$. By feasibility, it must be the case that x = 0.5 and $t_i^{\sigma} = x$.

(2) Consider agents $i \in N \setminus S_x^1$ and $j \in S_x^3$ such that $(i, j, t_i^{\sigma}) \in \sigma$. By Claim 1 and $j \in S_x^3$, we know $t_j^{\sigma} \ge 1 - x$. Agent *i* either belongs to S_x^3 or S_x^2 . If $i \in S_x^3$, then $t_i^{\sigma} \ge 1 - x$ (by Claim 1). Also since $j \in S_x^3$, $t_j^{\sigma} \ge 1 - x$. Feasibility implies 1 - x = 0.5. Thus $t_i^{\sigma} = 0.5 = x$ and $t_j^{\sigma} = 1 - x$. If $i \in S_x^2$, Claim 1 implies $x \le t_i^{\sigma} \le 1 - x$. We know $t_j^{\sigma} \ge 1 - x$. Feasibility implies $t_i^{\sigma} \ge 1 - x$.

Claim 6: Consider the allocation σ . If $x \neq 0.5$, we have

- 1. There does not exist a triple $(i, j, t_i^{\sigma}) \in \sigma$ such that $i, j \in S_x^1$.
- 2. There does not exist a triple $(i, j, t_i^{\sigma}) \in \sigma$ such that $i, j \in S_x^3$.

Proof: Part 1 follows immediately from Claim 1 and the definition of S_x^1 . Similarly Part 2 follows from Claim 1 and the definition of S_x^3 .

Claim 7: Consider agents $i, j \in S_x^2$ such that $p_i, p_j \in (x, 1-x)$. Then in the allocation σ , we have $\neg [t_i^{\sigma} = x \text{ and } t_j^{\sigma} = 1-x]$.

Proof: We assume for contradiction that there exist $i, j \in S_x^2$ such that $p_i, p_j \in (x, 1 - x)$, $t_i^{\sigma} = x$ and $t_j^{\sigma} = 1 - x$. Here $x < \min\{p_i, 1 - p_j\}$. Thus there exists $\epsilon > 0$ such that $x + \epsilon \succ_i t_i^{\sigma}$ and $1 - x - \epsilon \succ_j t_j^{\sigma}$. The pair (i, j) blocks σ . This is a contradiction as σ is stable by assumption.

We will first prove Part (a) of the lemma. Our aim is to calculate the cardinality of the set $\{i \in N : t_i^{\sigma} = x\}$. In order to do this, we will first deduce how agents are matched across the three groups. There are two cases to consider depending on the cardinalities of the sets S_x^1 and S_x^3 : (I) $|S_x^1| \ge |S_x^3|$ and (II) $|S_x^1| < |S_x^3|$.

(I) Consider the first case where $|S_x^1| \ge |S_x^3|$. We claim that in this case, all agents in S_x^3 are matched to agents in S_x^1 in σ . We assume for contradiction that there exists $i \in S_x^3$ who is not matched to an agent in S_x^1 in σ . By Claim 5 (Part 2), we know $t_i^{\sigma} = 1 - x$. Also there exists $j \in S_x^1$ such that j is not matched to an agent in S_x^3 . This is because $|S_x^1| \ge |S_x^3|$ and the assumption that an agent in S_x^3 is not matched to an agent in S_x^1 . By Claim 5 (Part 1), we know $t_j^{\sigma} = x$. We have shown that there exists $i \in S_x^3$ with $t_i^{\sigma} = 1 - x$ and $j \in S_x^1$ with $t_j^{\sigma} = x$. This is not possible by Claim 4.

Also for any $i \in S_x^3$, we have $t_i^{\sigma} \neq 1 - x$. Suppose not, i.e. $t_i^{\sigma} = 1 - x$. We have shown above that agent *i* is matched with some agent $j \in S_x^1$. Thus $t_j^{\sigma} = x$. This is not possible by Claim 4.

Now we will calculate the cardinality of the set $\{i \in N : t_i^{\sigma} = x\}$. There are two possibilities to consider.

1. If x = 0.5, then

$$|\{i \in N : t_i^{\sigma} = x\}| = |S_x^1| + |S_x^2| - |S_x^3|.$$

We have shown that all agents in S_x^3 are matched to agents in S_x^1 . Thus $|S_x^3|$ agents in S_x^1 are matched to agents in S_x^3 . Also for all $i \in S_x^3$, we have $t_i^{\sigma} \neq 1-x$. Thus the agents in S_x^1 matched to agents in S_x^3 do not get a contribution of x. The remaining agents in S_x^1 (the cardinality of this set is $|S_x^1| - |S_x^3|$) are matched to agents in $N \setminus S_x^3$. By Claim 5 (Part 1), we know all such agents get a contribution of x. Thus $|S_x^1| - |S_x^3|$ agents in S_x^1 have a contribution of x in σ . Since x = 0.5, we know $p_i = 0.5$ for all $i \in S_x^2$. By Claim 1, we have $t_i^{\sigma} = 0.5$ for all $i \in S_x^2$. So all agents in S_x^2 get a contribution of x in σ . Thus $|S_x^1| - |S_x^3|$ agents in S_x^3 receive x in σ .

2. If $x \neq 0.5$, then

$$|\{i \in N | t_i^{\sigma} = x\}| = \max\{|S_x^1| - |S_x^3| + |\{i \in S_x^2 : p_i = x\}|, |\{j \in S_x^2 : p_j = 1 - x\}|\}.$$

Note that $|S_x^1| - |S_x^3|$ agents in S_x^1 receive x in σ . Consider the set S_x^2 . For all $i \in S_x^2$ with $p_i = x$, we have $t_i^{\sigma} = x$. This is because $t_i^{\sigma} \succeq_i x$ (recall σ satisfies Condition (1) in the antecedent of the lemma). Similarly for all $i \in S_x^2$ with $p_i = 1 - x$, we have $t_i^{\sigma} = 1 - x$. This is because $t_i^{\sigma} \succeq_i 1 - x$ by Condition (1) of the lemma. Note that their partners in σ get x (recall Observation 4). Also none of the agents in S_x^3 receive x in σ . This follows from Claim 1 (all agents in S_x^3 get at least 1 - x) and the fact that x < 0.5. Thus max $\{|S_x^1| - |S_x^3| + |\{i \in S_x^2 : p_i = x\}|, |\{j \in S_x^2 : p_j = 1 - x\}|\}$ is the minimum number of agents in σ who receive x. We have,

$$|\{i \in N : t_i^{\sigma} = x\}| \ge \max\{|S_x^1| - |S_x^3| + |\{i \in S_x^2 : p_i = x\}|, |\{j \in S_x^2 : p_j = 1 - x\}|\}.$$

Suppose the above condition holds with strict inequality. We know that none of the agents in S_x^3 get x. Also exactly $|S_x^1| - |S_x^3|$ in S_x^1 receive x. Thus in the case of strict inequality, there exists an agent $i \in S_x^2$ with $p_i \in (x, 1 - x)$ and $t_i^{\sigma} = x$.

Note that $|\{i \in N : t_i^{\sigma} = x\}| = \{j \in N : t_j^{\sigma} = 1 - x\}$. Since the condition holds with strict inequality, we have

$$|\{j \in N : t_i^{\sigma} = 1 - x\}| > |\{j \in S_x^2 : p_j = 1 - x\}|.$$

Thus there exists $j \in N$ such that $p_j \neq 1 - x$ and $t_j^{\sigma} = 1 - x$. Note that j cannot belong to S_x^3 (recall we have shown that all agents in S_x^3 do not receive 1 - x). Also all agents in S_x^1 get at most x. So $j \in S_x^{2,28}$

So there exists $i, j \in S_x^2$ such that $p_i, p_j \in (x, 1-x)$, $t_i^{\sigma} = x$ and $t_j^{\sigma} = 1 - x$. This is not possible by Claim 7.

Thus,

$$|\{i \in N : t_i^{\sigma} = x\}| = \max\{|S_x^1| - |S_x^3| + |\{i \in S_x^2 : p_i = x\}|, |\{j \in S_x^2 : p_j = 1 - x\}|\}.$$

(II) The second case is $|S_x^1| < |S_x^3|$. We claim that all agents in S_x^1 are matched to agents in S_x^3 in σ . Also for all $i \in S_x^1$, we have $t_i^{\sigma} \neq x$.

In order to compute the cardinality of the set $\{i \in N | t_i^{\sigma} = x\}$, we compute the minimum number of agents that must receive 1 - x in σ .²⁹ There are two possibilities.

1. If x = 0.5, then

$$\{i \in N : t_i^{\sigma} = x\} = |S_x^3| - |S_x^1| + |S_x^2|.$$

²⁸Note that $p_j \neq x$. We have shown above that if $j \in S_x^2$ and $p_j = x$, then $t_j^{\sigma} = x$.

²⁹This number is useful as for each agent who gets 1 - x in σ , her partner gets x in σ (Observation 4).

All agents in S_x^1 are matched to agents in S_x^3 . Also none of the agents in S_x^1 get 1 - x in σ .³⁰ So $|S_x^1|$ agents in S_x^3 do not receive x in σ . By Claim 5 (Part 2), the remaining agents in S_x^3 (who are matched to agents in $N \setminus S_x^1$) must get 1 - x in σ . Thus $|S_x^3| - |S_x^1|$ agents in S_x^3 have a contribution of 1 - x in σ .

Since x = 0.5, we have $p_i = 0.5$ for all $i \in S_x^2$. By Claim 2, we know that all agents in S_x^2 have a contribution of x = 1 - x = 0.5 in σ .

Thus $|S_x^3| - |S_x^1|$ agents in S_x^1 , all agents in S_x^2 and none of the agents in S_x^1 receive 1 - x in σ .

2. If $x \neq 0.5$, then

$$|\{i \in N | t_i^{\sigma} = x\}| = \max\{|S_x^3| - |S_x^1| + |\{i \in S_x^2 : p_i = 1 - x\}|, |\{i \in S_x^2 : p_i = x\}|\}$$

We can prove this case using similar arguments to Part 2 in Case I.

We have shown that $|\{i \in N | t_i^{\sigma} = x\}|$ only depends on the preference profile \succ and x. In particular, it does not depend on the allocation σ .

Note that the cardinality of the set $\{i \in N : t_i^{\tau} = x\}$ can be computed in exactly the same manner as we did for the allocation σ .³¹ Thus we conclude $|\{i \in N : t_i^{\sigma} = x\}| = |\{i \in N : t_i^{\tau} = x\}|$.

We now prove Part (b) of the lemma. Consider agents $i, j \in N$ such that $p_i < 0.5$ and $p_j > 0.5$. There are two possibilities based on the cardinalities of $S_{0.5}^1$ and $S_{0.5}^3$.

In Case I where $|S_{0.5}^1| \ge |S_{0.5}^3|$, we have shown above that all agents in $S_{0.5}^3$ are matched to agents in $S_{0.5}^1$. Also each agent in $S_{0.5}^3$ has a contribution strictly greater than 0.5 (by Claim 2 (b)) in both σ and τ . Since $p_j > 0.5$, we know $j \in S_{0.5}^3$. Thus $t_j^{\tau} = 0.5$ is not possible.

In Case II where $|S_{0.5}^1| < |S_{0.5}^3|$, all agents in $S_{0.5}^1$ are matched to agents in $S_{0.5}^3$. Each agent in $S_{0.5}^1$ has a contribution strictly smaller than 0.5 (by Claim 2 (b)) in both σ and τ . Since $p_i < 0.5$, we know $i \in S_{0.5}^1$. Thus it is not possible that $t_i^{\sigma} = 0.5$.

OBSERVATION 5 Consider a preference profile \succeq and allocations σ and τ . If τ Pareto dominates σ at \succeq , then $I_{i,t_{\tau}^{\tau}} \subseteq I_{i,t_{\sigma}^{\sigma}}$ for all $i \in N$. This implies $D^{\tau} \subseteq D^{\sigma}$ and $U^{\tau} \subseteq D^{\sigma}$.

LEMMA 6 Consider a preference profile \succeq and allocations σ and τ . If σ satisfies Condition S and τ Pareto dominates σ , then τ satisfies Condition S.

³⁰This is because none of the agents in S_x^1 get x = 0.5.

³¹All the claims proved are true for the allocation τ as well. The only change required in the arguments is to replace Condition (1) of the lemma by Condition (2).

Proof: We assume for contradiction that τ does not satisfy Condition S. There are two cases. The first case is when τ violates Part 1 of Condition S, i.e. there exists $l \in L, h \in H$ such that $I_{h,t_h^{\tau}} \cap I_{l,t_l^{\tau}} \neq \emptyset$. Since $I_{l,t_l^{\tau}} \subseteq I_{l,t_l^{\sigma}}$ and $I_{h,t_h^{\tau}} \subseteq I_{h,t_h^{\sigma}}$ (by Observation 5), we have $I_{l,t_l^{\sigma}} \cap I_{h,t_h^{\sigma}} \neq \emptyset$. Thus σ violates Condition S. The second case is when τ violates Part 2 of Condition S, i.e. there exists $i \in N$ such that $0.5 \in I_{i,t_i^{\tau}}$. By Observation 5, $0.5 \in I_{i,t_i^{\sigma}}$. Thus σ violates Condition S.

OBSERVATION 6 Consider an allocation σ which satisfies Condition S. For any $x \in [0, 0.5]$, if $x \notin D^{\sigma} \cup U^{\sigma}$, then $[t_i^{\sigma} \succeq_i x \text{ and } t_i^{\sigma} \succeq_i 1 - x]$ for all $i \in N$.

LEMMA 7 Consider a preference profile \succeq . If σ satisfies Condition S and τ Pareto dominates σ at \succeq , then for all $x \in [0, 1]$,

$$|\{i \in N | t_i^{\tau} = x\}| = |\{i \in N | t_i^{\sigma} = x\}|.$$

Proof: Consider a preference profile \succeq and allocations σ and τ . Assume σ satisfies Condition S and τ Pareto dominates σ at \succeq . Consider an $x \in [0, 1]$. There are two possibilities.

(A) Let $x \in [0, 0.5]$. We consider three subcases.

1. $x \in [0, 0.5] \setminus [U^{\sigma} \cup D^{\sigma}].$

Since τ Pareto dominates σ , we have $U^{\tau} \subseteq U^{\sigma}$ and $D^{\tau} \subseteq D^{\sigma}$ (Observation 5). Thus $x \in [0, 0.5] \setminus [U^{\tau} \cup D^{\tau}]$. We know $x \notin D^{\sigma} \cup U^{\sigma}$. Observation 6 implies $t_i^{\sigma} \succeq_i x$ and $t_i^{\sigma} \succeq_i 1 - x$ for all $i \in N$. Similarly for the allocation τ , we have $t_i^{\tau} \succeq_i x$ and $t_i^{\tau} \succeq_i 1 - x$ for all $i \in N$.

Applying Lemma 5, we have

$$|\{i \in N : t_i^{\sigma} = x\}| = |\{i \in N : t_i^{\tau} = x\}|.$$

2. $x \in U^{\sigma}$. Then there exists $h \in H$ such that $x \in I_{h,t_h^{\sigma}}$ and $1 - x \succ_h t_h^{\sigma}$. Since σ satisfies Condition S, we have $0.5 \notin U^{\sigma}$. Thus x < 0.5.

For any $i \in N$ with $p_i = x$, it must be the case that $t_i^{\sigma} = x$. If not, then (i, h) will block σ with (x, 1 - x).

Also for any $i \in N$ with $t_i^{\sigma} = x$, it must be the case that $p_i = x$. If not, the pair (i, h) will block σ .³² This implies

$$|\{i \in N : t_i^{\sigma} = x\}| = |\{i \in N : p_i = x\}|.$$

³²Suppose $p_i < x$. Then there exists $\epsilon > 0$ such that $p_i < x - \epsilon < x$ and $x - \epsilon \in I_{h,t_h^{\sigma}}$. The pair (i,h) blocks σ with $(x - \epsilon, 1 - x + \epsilon)$. We can give a similar argument for the case $p_i > x$.

Note that all these agents must receive their peaks in τ as τ Pareto dominates σ . Thus,

$$|\{i \in N : t_i^{\tau} = x\}| \ge |\{i \in N : p_i = x\}| = |\{i \in N : t_i^{\sigma} = x\}|$$

3. $x \in D^{\sigma}$. Then there exists $l \in L$ such that $x \in I_{l,t_{\tau}^{\sigma}}$.

We will look at the agents who receive 1 - x in σ . For all $i \in N$ such that $p_i^{\sigma} = 1 - x$, we have $t_i^{\sigma} = 1 - x$. Also for all $i \in N$ with $t_i^{\sigma} = 1 - x$, it must be the case that $p_i = 1 - x$. Thus,

$$|\{i \in N : t_i^{\sigma} = 1 - x\}| = |\{i \in N : p_i = 1 - x\}|.$$

Since τ Pareto dominates σ , we have

$$|\{i \in N : t_i^{\tau} = 1 - x\}| \ge |\{i \in N : p_i = 1 - x\}| = |\{i \in N : t_i^{\sigma} = 1 - x\}|.$$

Note that $|\{i \in N : t_i^{\tau} = 1 - x\}| = |\{i \in N : t_i^{\tau} = x\}|$ and $|\{i \in N : t_i^{\sigma} = 1 - x\}| = |\{i \in N : t_i^{\sigma} = x\}|$ (Observation 4).

Therefore,

$$|\{i \in N : t_i^{\tau} = x\}| \ge |\{i \in N | p_i = 1 - x\}| = |\{i \in N | t_i^{\sigma} = 1 - x\}| = |\{i \in N | t_i^{\sigma} = x\}|.$$

We have shown that for every $x \in [0, 0.5]$,

$$|\{i \in N : t_i^{\tau} = x\}| \ge |\{i \in N : t_i^{\sigma} = x\}|.$$

(B) $x \in (0.5, 1]$.

Note that $|\{i \in N : t_i^{\tau} = x\}| = |\{i \in N : t_i^{\tau} = 1 - x\}|$ and $|\{i \in N : t_i^{\sigma} = x\}| = |\{i \in N : t_i^{\sigma} = x\}|$ (by Observation 4). Since $1 - x \in [0, 0.5]$, Case (A) is applicable. So $|\{i \in N : t_i^{\tau} = 1 - x\}| \ge |\{i \in N : t_i^{\sigma} = 1 - x\}|$. Combining the facts above, we have $|\{i \in N : t_i^{\tau} = x\}| \ge |\{i \in N | t_i^{\sigma} = x\}|$.

From (A) and (B), we know

$$|\{i \in N : t_i^{\tau} = x\}| \ge |\{i \in N | t_i^{\sigma} = x\}| \text{ for all } x \in [0, 1]$$
(2)

The sum of both the LHS and RHS of Equation 2 over all $x \in [0, 1]$ is |N|. Suppose there exists some x for which Equation 2 holds with strict inequality. Then there must exist another x for which the LHS of Equation 2 will be strictly less than the RHS. This contradicts 2. Thus Equation 2 holds with equality.

This completes the proof of the lemma.

OBSERVATION 7 Consider an allocation σ that satisfies Condition S. For any agent $i \in N$, we have (a) if $t_i^{\sigma} < 0.5$ then $p_i < 0.5$ and (b) if $t_i^{\sigma} > 0.5$ then $p_i > 0.5$. To prove (a), assume $t_i^{\sigma} < 0.5$ and $p_i \ge 0.5$ for some agent *i*. Then 0.5 belongs to $I_{i,t_i^{\sigma}}$ and σ violates Condition S. Similarly we can prove (b). Also if $p_i = 0.5$ then $t_i^{\sigma} = 0.5$.

DEFINITION 5 Consider a preference profile \succeq and two allocations $\sigma, \tau \in \Sigma$. Assume τ Pareto dominates σ at \succeq . We say τ is minimal if for all $\gamma \in \Sigma$ such that γ Pareto dominates σ at \succeq , we have

$$|\{i \in N : t_i^{\tau} \neq t_i^{\sigma}\}| \le |\{i \in N : t_i^{\gamma} \neq t_i^{\sigma}\}|.$$

OBSERVATION 8 Consider a preference profile \succeq and allocations $\sigma, \gamma \in \Sigma$ such that γ Pareto dominates σ at \succeq . Since |N| is finite, there exists an allocation $\tau \in \Sigma$ such that τ Pareto dominates σ at \succeq and is minimal. This allocation may not be unique.

We now complete the proof of Pareto efficiency.

Proof: Let σ be the allocation generated by the algorithm. We will prove the theorem by contradiction. Suppose there exists an allocation $\gamma \in \Sigma$ such that γ Pareto dominates σ . By Observation 8, we know there exists $\tau \in \Sigma$ such that τ Pareto dominates σ and is minimal. We will now work with the allocation τ and use it to show a contradiction.

Note that σ satisfies Condition S as it is generated by the algorithm. Since τ Pareto dominates σ , we know τ also satisfies Condition S (Lemma 6).

There are two cases to consider: (I) there exists an agent $l \in L$ such that $t_l^{\tau} \neq t_l^{\sigma}$ and (II) $t_l^{\tau} = t_l^{\sigma}$ for all $l \in L$.

Case I: There exists an agent $l \in L$ such that $t_l^{\tau} \neq t_l^{\sigma}$.

The SAM algorithm generates σ in K steps.³³ There are two possibilities. The first possibility is that there exists a low type agent l who is matched in Step 0 and $t_l^{\sigma} \neq t_l^{\tau}$.³⁴ We denote agent l as $l_{\bar{i}}$ where $\bar{i} = 0$. The second possibility is that all low type agents matched in Step 0 (if any) have the same contribution values in both σ and τ . Then there exists a low type agent who is matched in some Step $q \ge 1$ and has different contribution values in σ and τ . Let $\bar{i} \in \{1, \ldots, K\}$ be the smallest integer such that (i) $t_{l_{\bar{i}}}^{\sigma} \neq t_{l_{\bar{i}}}^{\tau}$ and (ii) $t_{l_i}^{\sigma} = t_{l_i}^{\tau}$ for all $i < \bar{i}$. Note that $l_{\bar{i}}$ is the agent matched in Step \bar{i} of the algorithm.

There are three cases based on the contribution value of agent $l_{\bar{i}}$ in σ .

³³Recall in each step q (where $1 \le q \le K$), a low type agent l_q is matched. In Step 0, low type agents are matched if |L| > |H|.

³⁴If there are several such agents, we choose an agent arbitrarily from this set.

1. $t_{l_{\bar{i}}}^{\sigma} = p_{l_{\bar{i}}}$.

Since τ Pareto dominates σ , we have $t_{l_i}^{\tau} = p_{l_i}$. This is a contradiction as by assumption the contribution values of agent l_i are different in σ and τ .

2. $t_{l_{\bar{i}}}^{\sigma} < p_{l_{\bar{i}}}$.

Here $t_{l_{\bar{i}}}^{\sigma}$ is the infimum of the improvement set of $l_{\bar{i}}$ in σ . Since τ Pareto dominates σ and $t_{l_{\bar{i}}}^{\sigma} \neq t_{l_{\bar{i}}}^{\tau}$, we have $t_{l_{\bar{i}}}^{\tau} \in \overline{I_{l_{\bar{i}}, t_{l_{\bar{i}}}^{\sigma}}} \setminus \{t_{l_{\bar{i}}}^{\sigma}\}$. Note that $t_{l_{\bar{i}}}^{\sigma} < t_{l_{\bar{i}}}^{\tau} \leq 0.5$.³⁵

Let $t_{l_i}^{\tau} = m^*$. Since σ satisfies Condition S and τ Pareto dominates it, applying Lemma 7 at $x = m^*$, we have

$$|\{i \in N : t_i^{\sigma} = m^*\}| = |\{i \in N | t_i^{\tau} = m^*\}|.$$

By assumption, $l_{\bar{i}}$ does not belong to the former set but belongs to the latter. Thus there exists an agent j who belongs to the former set and not to the latter. Note that $t_j^{\sigma} = m^* \neq t_j^{\tau}$.

We claim that $j \in L$. To show this, we consider two cases. The first case is when $m^* < 0.5$. Suppose $j \notin L$. Then $p_j \ge 0.5$ and $0.5 \in I_{j,t_j^{\sigma}}$. This results in a contradiction as σ satisfies Condition S. Thus when $m^* < 0.5$, it must be the case that $p_j < 0.5$ and $j \in L$. The remaining case is when $m^* = 0.5$. We know $t_{l_i}^{\tau} = m^* = 0.5$, $p_{l_i} < 0.5$ and $t_j^{\sigma} = m^* = 0.5$. Applying Lemma 5 (Part (b)), we have $p_j \le 0.5$. We will show that $p_j \ne 0.5$ if τ satisfies Condition S. Suppose $p_j = 0.5$. We know $t_j^{\tau} \ne t_j^{\sigma} = m^* = 0.5$. Then $p_j = 0.5 \in I_{j,t_i^{\tau}}$ and τ violates Condition S. Thus $p_j < 0.5$ and $j \in L$.

We know $l_{\bar{i}}, j \in L$. Agent j has different contribution values in σ and τ . By assumption, $l_{\bar{i}}$ is the first low type agent who has different contribution values in σ and τ . We first argue that $\bar{i} \neq 0$. If $\bar{i} = 0$, we have $t_{l_{\bar{i}}}^{\sigma} = 0.5$. Since $p_{l_{\bar{i}}} < 0.5$, we have a contradiction to the assumption $t_{l_{\bar{i}}}^{\sigma} < p_{l_{\bar{i}}}$. Thus $\bar{i} \geq 1$ and agent j is matched in a step strictly greater than Step \bar{i} in the algorithm. By Lemma 4, we have $t_{j}^{\sigma} = m^* \leq t_{l_{\bar{i}}}^{\sigma}$. We have a contradiction as $t_{l_{\bar{i}}}^{\sigma} < t_{j}^{\tau} = m^* = t_{j}^{\sigma}$.

3. $t_{l_{\bar{i}}}^{\sigma} > p_{l_{\bar{i}}}$.

Let $t_{l_i}^{\sigma} = s^*$. So $p_{l_i} < t_{l_i}^{\sigma} = s^* \leq 0.5$ (by Observation 7 and the fact that σ satisfies Condition S).

Since σ satisfies Condition S and τ Pareto dominates it, applying Lemma 7 at $x = s^*$, we have

$$|\{i \in N : t_i^{\sigma} = s^*\}| = |\{i \in N : t_i^{\tau} = s^*\}|.$$

³⁵We know τ satisfies Condition S. By Observation 7 and $p_{l_{\bar{i}}} < 0.5$, we have $t_{l_{\bar{i}}}^{\tau} \leq 0.5$.

Agent $l_{\bar{i}}$ belongs to the first set and not to the latter. Thus there exists an agent j who belongs to the latter set and not to the first. Note that $t_j^{\tau} = s^* = t_{l_{\bar{i}}}^{\sigma} \neq t_j^{\sigma}$.

We claim that $j \in L$. To show this, we consider two cases. The first case is when $s^* < 0.5$. Suppose $j \notin L$ and $p_j \ge 0.5$. Then $0.5 \in I_{j,t_j^{\tau}}$ and τ violates Condition S (Observation 7). Hence $p_j < 0.5$ and $j \in L$. The second case is when $s^* = 0.5$. We know $t_{l_i}^{\sigma} = s^* = 0.5$, $p_{l_i} < 0.5$ and $t_j^{\tau} = s^* = 0.5$. Applying Lemma 5 (Part (b)), we have $p_j \le 0.5$. We will show that $p_j \neq 0.5$ if σ satisfies Condition S. Suppose $p_j = 0.5$. We know $t_j^{\sigma} \neq s^* = 0.5$. Then $p_j = 0.5 \in I_{j,t_j^{\sigma}}$ and σ violates Condition S. Thus $p_j < 0.5$ and $j \in L$. This establishes the claim that $j \in L$.

We now show that $e_j(s^*) \leq e_{l_i}(s^*)$. To prove this claim, we consider two cases.

The first case is when $\overline{i} \geq 1$. Recall that both agents $l_{\overline{i}}$ and j have different contribution values in σ and τ . Since $\overline{i} \geq 1$ and agent $l_{\overline{i}}$ is the first low type agent to have different contribution values in σ and τ , agent j is matched in a step strictly greater than Step \overline{i} . Thus $j \in L_{\overline{i}}$.

Since $p_{l_{\bar{i}}} < t^{\sigma}_{l_{\bar{i}}}$, we know this is only possible in Substep $\bar{i}.1$ or Substep $\bar{i}.3.1$ of the algorithm (by Lemma 2). In both cases, $l_{\bar{i}}$ is the secondary agent. Thus $e_l(s^*) \leq e_{l_{\bar{i}}}(s^*)$ for all $l \in L_{\bar{i}} \setminus \{l_{\bar{i}}\}$. Since $j \in L_{\bar{i}}$, we have $e_j(s^*) \leq e_{l_{\bar{i}}}(s^*)$.

The second case is when $\overline{i} = 0$. Note that $t_{l_{\overline{i}}}^{\sigma} = s^* = 0.5 \neq t_j^{\sigma}$. This means that j is not matched in Step 0, when $l_{\overline{i}}$ is matched. Thus $e_j(0.5) \leq e_{l_{\overline{i}}}(0.5)$. This completes the proof of the claim.

Thus,

$$\overline{I_{l_{\bar{i}},t_{l_{\bar{i}}}^{\sigma}}} = [e_{l_{\bar{i}}}(s^*), s^*] \cap [0, 0.5] \subseteq [e_j(s^*), s^*] \cap [0, 0.5] = \overline{I_{j,t_j^{\tau}}}$$
(3)

$$I_{l_{\bar{i}},t_{l_{\bar{i}}}^{\sigma}} = (e_{l_{\bar{i}}}(s^*), s^*) \cap [0, 0.5] \subseteq (e_j(s^*), s^*) \cap [0, 0.5] = I_{j,t_j^{\tau}}$$
(4)

Since τ Pareto dominates σ , we have $t_{l_{\bar{i}}}^{\tau} \in \overline{I_{l_{\bar{i}}, t_{l_{\bar{i}}}^{\sigma}}}$. This together with Fact 3 implies $t_{l_{\bar{i}}}^{\tau} \in \overline{I_{j, t_{j}}}$. So $t_{l_{\bar{i}}}^{\tau} \succeq_{j} t_{j}^{\sigma}$. As τ Pareto dominates σ , we know $t_{j}^{\tau} \succeq_{j} t_{j}^{\sigma}$. Hence for agent j,

$$t_{l_{\tilde{i}}}^{\tau} \gtrsim_{j} t_{j}^{\tau} \gtrsim_{j} t_{j}^{\sigma} \tag{5}$$

Note that if $t_{l_{\bar{i}}}^{\tau} \in I_{l_{\bar{i}}, t_{l_{\bar{i}}}^{\sigma}}$, then $t_{l_{\bar{i}}}^{\tau} \in I_{j, t_{j}}^{\tau}$ (by Fact 4). Thus $t_{l_{\bar{i}}}^{\tau} \succ_{j} t_{j}^{\tau} \succeq_{j} t_{j}^{\sigma}$ (as τ Pareto dominates σ). This implies $t_{l_{\bar{i}}}^{\tau} \succ_{j} t_{j}^{\sigma}$. Thus

If
$$t_{l_{\bar{i}}}^{\tau} \succ_{l_{\bar{i}}} t_{l_{\bar{j}}}^{\sigma}$$
 then $t_{l_{\bar{i}}}^{\tau} \succ_{j} t_{j}^{\sigma}$ (6)

We will now construct an allocation $\delta \in \Sigma$ such that

(a) δ Pareto dominates σ and

(b) $|\{i \in N : t_i^{\delta} \neq t_i^{\sigma}\}| < |\{i \in N : t_i^{\tau} \neq t_i^{\sigma}\}|.$

This will contradict the assumption that τ Pareto dominates σ and is minimal.

We construct δ as follows. The pairs in δ are defined as,

- The partner of $l_{\overline{i}}$ in δ is the partner of j in τ .
- The partner of j in δ is the partner of $l_{\overline{i}}$ in τ .
- For any agent s (different from $l_{\bar{i}}, j$ and their partners' in τ), the partner of s in δ is the same as her partner in τ .

To obtain the pairs in δ , we interchange the partners of agents $l_{\bar{i}}$ and j in τ and the partners of all other agents remain unchanged.

The contributions of the agents in δ are defined as,

- $t_s^{\delta} = t_s^{\tau}$ for all $s \in N \setminus \{l_i, j\}$. The contribution of all such agents in δ is equal to their contribution in τ .
- $t_{l_{\bar{i}}}^{\delta} = t_{l_{\bar{i}}}^{\sigma} = s^*$. The contribution of $l_{\bar{i}}$ in δ is equal to her contribution in σ .
- $t_j^{\delta} = t_{l_i}^{\tau}$. The contribution of agent j in δ is equal to the contribution of agent l_i in τ .

Note that for all $s \in N \setminus \{l_{\bar{i}}, j\}$, we have $t_s^{\delta} = t_s^{\tau} \succeq_s t_s^{\sigma}$ as τ Pareto dominates σ . The contribution of agent $l_{\bar{i}}$ is the same in δ and σ . For agent j, we have $t_j^{\delta} = t_{\bar{l}_{\bar{i}}}^{\tau} \succeq_j t_j^{\sigma}$ (Fact 5). Thus all agents in N weakly prefer their contributions in δ to their contributions in σ .

In order to show that δ Pareto dominates σ , we need to show that there exists an agent who strictly improves in δ with respect to σ .³⁶ We consider three cases based on the agent who strictly prefers her contribution in τ to that in σ . Let $k \in N$ be the agent who strictly improves from σ to τ .

- (a) $k \in N \setminus \{l_{\bar{i}}, j\}$. Here $t_k^{\delta} = t_k^{\tau} \succ_k t_k^{\sigma}$. Thus k also strictly improves in δ in comparison to σ .
- (b) $k = l_{\bar{i}}$. Here $t_{l_{\bar{i}}}^{\tau} \succ_{l_{\bar{i}}} t_{l_{\bar{i}}}^{\sigma}$. Then Fact 6 implies $t_j^{\delta} = t_{l_{\bar{i}}}^{\tau} \succ_j t_j^{\sigma}$. Thus j strictly improves in δ in comparison to σ .
- (c) k = j. Here $t_j^{\tau} \succ_j t_j^{\sigma}$. By Fact 5, $t_j^{\delta} = t_{l_{\bar{i}}}^{\tau} \succeq_j t_j^{\tau}$. Thus $t_j^{\delta} \succ_j t_j^{\sigma}$ and j strictly improves in δ in comparison to σ .

³⁶Note that this will also establish that σ and δ are distinct.

We have established that there exists an agent who strictly improves in δ with respect to σ .

We now show that τ is not minimal. Let K be the number of agents in $N \setminus \{l_{\bar{i}}, j\}$ who have different contribution values in σ and τ . By contruction, K is also the number of agents in $N \setminus \{l_{\bar{i}}, j\}$ who have different contribution values in σ and δ .

Note that $|\{i \in N : t_i^{\tau} \neq t_i^{\sigma}\}| = K + 2$ and $|\{i \in N : t_i^{\delta} \neq t_i^{\sigma}\}| \leq K + 1$. So τ is not minimal and we have a contradiction.

Case II: For all $l \in L$, $t_l^{\tau} = t_l^{\sigma}$ Then there exists $h \in H$ such that $t_h^{\tau} \neq t_h^{\sigma}$.

The proof of Case II is virtually identical to that of Case I. We omit the details.

This completes the proof of the theorem.

References

- BARBERÀ, S., M. JACKSON, AND A. NEME (1997): "Strategy-proof Allotment Rules," Games and Economic Behavior, 18, 1–21.
- BERGANTIÑOS, G., J. MASSÓ, I. M. DE BARREDA, AND A. NEME (2015): "Stable partitions in many division problems: the proportional and the sequential dictator solutions," Theory and Decision, 79, 227–250.
- BOCHET, O., R. ILKILIÇ, AND H. MOULIN (2013): "Egalitarianism under earmark constraints," Journal of Economic Theory, 148, 535–562.
- BOCHET, O., R. ILKILIÇ, H. MOULIN, AND J. SETHURAMAN (2012): "Balancing supply and demand under bilateral constraints," Theoretical Economics, 7, 395–423.
- CHING, S. (1994): "An alternative characterization of the uniform rule," <u>Social Choice and</u> Welfare, 11, 131–136.
- ERIKSSON, K. AND J. KARLANDER (2001): "Stable outcomes of the roommate game with transferable utility," International Journal of Game Theory, 29, 555–569.
- GALE, D. AND L. S. SHAPLEY (1962): "College admissions and the stability of marriage," The American Mathematical Monthly, 69, 9–15.
- GENSEMER, S., L. HONG, AND J. S. KELLY (1996): "Division rules and migration equilibria," Journal of Economic Theory, 69, 104–116.

- KLAUS, B., H. PETERS, AND T. STORCKEN (1998): "Strategy-proof division with singlepeaked preferences and individual endowments," Social Choice and Welfare, 15, 297–311.
- MAS-COLELL, A., M. D. WHINSTON, J. R. GREEN, ET AL. (1995): <u>Microeconomic</u> Theory, vol. 1, Oxford university press New York.
- MOULIN, H. (1980): "On strategy-proofness and single peakedness," <u>Public Choice</u>, 35, 437–455.
- NICOLÒ, A., A. SEN, AND S. YADAV (2019): "Matching with partners and projects," Journal of Economic Theory, 184, 104942.
- SCARF, H. E. (1967): "The core of an N person game," <u>Econometrica</u>: Journal of the Econometric Society, 50–69.
- SCHUMMER, J. AND W. THOMSON (1997): "Two derivations of the uniform rule and an application to bankruptcy," Economics Letters, 55, 333–337.
- SHAPLEY, L. AND R. VOHRA (1991): "On Kakutani's fixed point theorem, the KKMS theorem and the core of a balanced game," Economic Theory, 1, 108–116.
- SHAPLEY, L. S. AND M. SHUBIK (1971): "The assignment game I: The core," <u>International</u> Journal of Game Theory, 1, 111–130.
- SÖNMEZ, T. (1994): "Consistency, monotonicity, and the uniform rule," <u>Economics Letters</u>, 46, 229–235.
- SPRUMONT, Y. (1991): "The division problem with single-peaked preferences: a characterization of the uniform allocation rule," <u>Econometrica</u>: Journal of the Econometric Society, 509–519.
- TEO, C.-P. AND J. SETHURAMAN (1998): "The geometry of fractional stable matchings and its applications," Mathematics of Operations Research, 23, 874–891.